

Temi di Discussione

(Working Papers)

Public investment and monetary policy stance in the euro area

by Lorenzo Burlon, Alberto Locarno, Alessandro Notarpietro and Massimiliano Pisani

January 1150



Temi di discussione

(Working papers)

Public investment and monetary policy stance in the euro area

by Lorenzo Burlon, Alberto Locarno, Alessandro Notarpietro and Massimiliano Pisani

Number 1150 - December 2017

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Ines Buono, Marco Casiraghi, Valentina Aprigliano, Nicola Branzoli, Francesco Caprioli, Emanuele Ciani, Vincenzo Cuciniello, Davide Delle Monache, Giuseppe Ilardi, Andrea Linarello, Juho Taneli Makinen, Valerio Nispi Landi, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

PUBLIC INVESTMENT AND MONETARY POLICY STANCE IN THE EURO AREA

by Lorenzo Burlon*, Alberto Locarno*, Alessandro Notarpietro* and Massimiliano Pisani*

Abstract

This paper evaluates the macroeconomic impact of a programme for public infrastructure spending in the euro area (EA) under alternative assumptions about funding sources and the monetary policy stance. The quantitative assessment is made by simulating a dynamic general equilibrium model of a monetary union with region-specific fiscal policy. The main results are the following. First, EA-wide stimuli are more effective than unilateral (region-specific) stimuli. Second, under EA-wide stimulus, the fiscal multiplier is close to 2 if the forward guidance (FG) on the short-term policy rate holds. Third, if the monetary authority keeps down both the policy rates (with FG) and the long-term interest rates (with quantitative easing), the fiscal multiplier exceeds 3 at peak and investment spending is self-financing. Fourth, the financing method is relevant: debt financing, particularly under an accommodative monetary policy stance and if the sovereign spreads do not increase, is more growth-friendly than tax financing in the short-term (but not in the long-term). Fifth, the effectiveness of the fiscal stimulus is larger if government spending is directed towards productive goods and its implementation occurs efficiently and without delays.

JEL Classification: E52, E62, F41, F42.

Keywords: public investment, fiscal policy, monetary policy, euro area.

Contents

1. Introduction	
2. The model	9
2.1 Overview	
2.2 Restricted households	
2.3 Unrestricted households	
2.4 Capital producers	
2.5 Public capital and firms' decisions	15
2.6 Fiscal sector	
2.7 Monetary authority	
2.8 Equilibrium	
2.9 Calibration	
3. Simulated scenarios	
4. Results	
4.1 Home-country increase in public investment	
4.2 EA increase in public investment	
4.3 EA-wide public investment increase and non-standard monetary policy	measures:
4.4 EA-wide public investment financed with distortionary taxation	
4.5 Increase in sovereign spread	
4.6 EA-wide public investment under monetary accommodation	
5. Sensitivity analysis	
6. Conclusions	
References	
Tables and figures	39

* Bank of Italy, Directorate General for Economics, Statistics and Research.

Common action is urgently needed to sustain public investment in the euro area, which has fallen by a quarter in four years. Ignazio Visco, Governor of the Bank of Italy.¹

In parallel, it may be useful to have a discussion on the overall fiscal stance of the euro area with the view to raising public investment where there is fiscal space to do so. Mario Draghi, President of the European Central Bank.²

1 Introduction³

Weak aggregate demand and very low inflation have characterized the euro area (EA) economy since the Great Recession and notwithstanding an extremely accommodative monetary stance. In addition, by reducing the pace of capital accumulation and increasing the duration of unemployment, they have negatively affected potential output growth, contributing to make the recession to last longer and the recovery more subdued.

As proposed by policy-makers and commentators, one way to address these concerns and at the same time exploit the exceptionally low level of borrowing costs is to increase public spending in infrastructures.⁴ This measure is viewed as beneficial and effective for several reasons. First, it would be implemented after years of reductions in public investment spending and after a prolonged decline in the stock of public capital-to-GDP ratio.⁵ Second, it would stimulate aggregate demand in the short term and boost supply capacity in the long run. Third, it could be self-financing, if the impact on economic activity is large enough. The institutional and policy framework of the EA would provide an even more friendly environment for public infrastructure

¹See Visco (2014).

 $^{^{2}}$ See Draghi (2014).

 $^{^{3}}$ We thank Bartosz Mackowiak and participants at the Eurosystem Working Group on Econometric Modelling (May 2017), European Central Bank workshop "Euro area business investment in a global context – the role of cyclical and structural factors and frictions" (June 2017), Computing in Economics and Finance (June 2017), 5th Workshop in Macro Banking and Finance (September 2017). The views expressed in this paper are those of the authors alone and should not be attributed to the Bank of Italy or the Eurosystem. Any remaining errors are the sole responsibility of the authors.

⁴See International Monetary Fund (2014b), European Commission (2015), In 't Veld (2016), and OECD (2016). ⁵The International Monetary Fund, in Chapter 3 of the April 2014 World Economic Outlook, suggests that aging infrastructure and insufficient maintenance and investment may be affecting the quality of the existing infrastructure stock. See International Monetary Fund (2014a).

investment. Given that the fiscal policy is managed at the level of Member States, which mostly trade with each other, cross-country coordination in the form of simultaneous fiscal stimuli would favor bilateral exports (and, thus, limit the size of trade leakages associated with a fiscal stimulus implemented by a single Member State). Moreover, the stance of the common monetary policy, if expansionary as currently, would favour the crowding-in of households' and firms' spending, by capping the increase in short- and long-term interest rates that arises in correspondence of a fiscal stimulus financed by issuing public debt.

The ongoing debate usually acknowledges that there are also arguments against the use of infrastructure spending for demand management purposes. The most often quoted reasons are: (i) the lack of fiscal space, (ii) decision and implementation delays, (iii) disparities between the amounts spent and the actual value of the works realized, and (iv) distorted political incentives.⁶ First, the current debt levels are at historical highs in several countries and ambitious spending plans could backfire if debt sustainability concerns triggered an increase in sovereign risk premia. Second, the length of the political decision process and the need to comply with an increasingly complex regulatory environment make it extremely difficult to kick-start infrastructure projects quickly and have them completed in a reasonable amount of time. Infrastructure spending is therefore a poor way to try and manage aggregate demand. Third and fourth, most investment projects are selected not on the basis of a standard cost-benefit analysis, but rather with the aim of either favouring specific constituencies or in the hope of promoting the economic prospects of disadvantaged areas. If these areas are in a process of long-term economic decline, public investment will likely yield poor returns.

This paper evaluates the macroeconomic impact on the EA economy of the increase in public investment in infrastructures under alternative assumptions about the monetary policy response. The assessment is based on simulations of a calibrated three-country large-scale New Keynesian dynamic general equilibrium model, which includes two EA regions (one labelled "Home", calibrated to a relatively small-size economy, and the other assembling the rest of the eurozone, dubbed "REA") and the rest of the world ("RW"). The model is akin to the Eurosystem EAGLE (see Gomes *et al.*, 2010). The EA is a monetary union, so Home and REA share the same monetary policy and nominal exchange rate. The inclusion of the RW allows for a full characterization

⁶See Banca d'Italia (2012).

of trade flows. The model features country-specific fiscal policies; the public-sector budget exhibits, on the expenditure side, lump-sum transfers, public consumption, and public investment in infrastructures. On the revenue side, there are distortionary taxes on labor income, capital income, and consumption. Public debt is stabilized through a fiscal rule adjusting lump-sum transfers to achieve the desired debt target. Public capital enters the production function of domestic goods jointly with private capital and labor. As the speed at which spending occurs is crucial for short-run stimulative effects, we also allow for delays in the implementation of spending plans in infrastructures (time-to-build assumption). The monetary authority can resort to both standard and non-standard monetary policies, the latter including forward guidance (FG) and quantitative easing (QE).

The second novel feature of the model is financial segmentation à la Chen *et al.* (2012), which allows us to relax the well-known "Wallace neutrality" and make financial assets imperfect substitute so that sovereign bonds' purchases by the monetary authority have real effects in our model. In each EA region there are two types of households, "restricted" and "unrestricted". Restricted households can invest only in domestic long-term sovereign bond market and, because they are owner of domestic private capital producers (joint with domestic unrestricted households), in physical capital. The purchase of long-term government bonds by the monetary authority reduces long-term interest rates and therefore induces restricted households to increase consumption and investment via the standard intertemporal substitution effect.

In all exercises, we simulate an increase in public investment spending equal to 1% of beforeshock (steady-state) GDP. The fiscal stimulus lasts for five years and thereafter gradually fades out.

We initially consider the case of an increase in public investment implemented by the Home country in isolation under standard monetary policy (i.e., the short-term interest rate is set according to a Taylor rule that reacts to EA-wide inflation and output growth); we also consider a scenario in which the EA monetary authority promises to keep, in a credible way, the shortterm interest rate at its baseline level for the initial eight quarters ("Forward Guidance" on the monetary policy rate, FG). Public investment is financed by issuing government bonds; the fiscal rule ensuring debt stabilisation becomes active after five years from the beginning of the simulation.

We consider several variants, among which is the case of an EA-wide increase in public investment (i.e., an increase that takes place simultaneously in both the home country and in the rest of the monetary union) under both a standard monetary response and, alternatively, under the combination of (1) two-year FG and (2) QE. The latter aims at stimulating economic activity by lowering the interest rates on EA long-term sovereign bonds and its size is set equal to that of the public sector purchase programme (PSPP) announced by the ECB in January 2015, consisting in monthly purchases of $\in 60$ billion for nineteen months (i.e., covering seven quarters in our quarterly model). To highlight the relevance of the interaction between fiscal and monetary policies, we also evaluate the effectiveness of the stimulus when the public investment increase is, alternatively, (i) financed by issuing debt and, simultaneously, the sovereign spread increases, (ii) financed by distortionary taxation, and (iii) matched with a corresponding increase in the balance sheet of the monetary authority.

In the final section we present the outcome of the sensitivity analysis. We report how results change under alternative assumptions on (i) the efficiency of public spending (public spending is pure waste and does not contribute to public capital accumulation), (ii) the implementation delays associated with the administrative procedures for planning, bidding, and contracting (time-to-build assumption).⁷

All simulations are run under the assumption of perfect foresight. Accordingly, (i) the increase in public investment and the monetary policy stance are fully credible; (ii) there is no uncertainty, and (iii) households and firms correctly anticipate the paths of the policy variables.

The main results are the following. First, EA-wide stimuli are more effective than unilateral (region-specific) stimuli. Second, under EA-wide stimulus, the fiscal multiplier is close to 2 if the forward guidance (FG) on the short-term policy rate holds. Third, if the monetary authority keeps lower both the policy rates (with FG) and the long-term interest rates (with quantitative easing), the fiscal multiplier exceeds 3 at peak and investment spending is self-financing. Fourth, the financing method is relevant. Debt financing, in particular under an accommodative

⁷See Kydland and Prescott (1982) and, more recently, Bouakez et al. (2017).

monetary stance and if the sovereign spreads do not increase, is more growth-friendly than tax financing in the short run (but not in the long run).⁸ Fifth, the effectiveness of the fiscal stimulus is larger if government spending is directed towards productive goods and its implementation occurs efficiently and without delays.

Our paper is related to several contributions in the literature. Coenen *et al.* (2013) evaluates the impact on EA GDP of the European Economic Recovery Plan (EERP), enacted in response to the financial crisis of 2008–2009 and including, among the several measures, public investment. Elekdag and Muir (2014) evaluates the macroeconomic impact of public investment in infrastructure in the EA and in Germany. Blanchard *et al.* (2014) assesses the effects of fiscal expansions in European core economies on GDP in non-core countries. Abiad *et al.* (2016) stress that the effectiveness of public investment depends on its efficiency and on whether it is financed by debt issuance. De Jong *et al.* (2017) analyse the sensitivity of the effect of an increase in public investment in the EU to alternative assumptions regarding the structure of the economy and other policy provisions. Differently from the mentioned contributions, we focus on the interaction between fiscal policy and standard and non-standard monetary policy measures.

The paper is organized as follows. Section 2 reports the key equations and the calibration of the model. Section 3 illustrates the design of the exercises. Section 4 shows the results of the simulations. Section 5 presents evidence coming from the sensitivity analysis. Section 6 concludes.

2 The model

We first report an overview of the model. Subsequently, we describe its key features, i.e., households, capital producers, public capital and public investment, the fiscal authority budget constraint and fiscal rule, the EA monetary authority, and the definition of general equilibrium. Finally, we show the calibration.

 $^{^{8}}$ See International Monetary Fund (2015) for potentially adverse effects of debt-financing when sovereign risk-premia respond to announcements of fiscal stimulus, especially for countries with a high level of initial debt.

2.1 Overview

The model is New Keynesian and represents a world economy composed of three regions, i.e., Home, REA (Home+REA=EA), and RW. The size of the world economy is normalized to 1. Home, REA, and RW have sizes equal to n, n^* , and $(1 - n - n^*)$, with n > 0, $n^* > 0$, and $n + n^* < 1.^9$ Home and REA share the currency and the monetary authority. The latter sets the nominal interest rate according to EA-wide variables (a standard Taylor rule holds) when it does not deliberately enact non-standard monetary policy measures. The presence of the RW outside the EA allows to assess the role of the nominal exchange rate and extra-EA trade for the transmission of the EA shocks.

The first crucial feature of the model is that the EA firms can sell the final investment goods not only to domestic households but also to the domestic public sector. The former exploits investment goods to accumulate "private" physical capital, the latter to accumulate "public" capital. The public investment decision is an exogenous variable set by the fiscal authority. Crucially, domestic public capital enters the production of intermediate tradable and non-tradable goods joint with capital and labor supplied by domestic households. Public capital is common to both sectors. Firms take public capital as given when choosing their optimal demand for private capital and labor. The public capital does not provide any pecuniary return, but increases productivity of the private inputs and, thus, their returns.

The second novel feature of the model is financial segmentation à la Chen *et al.* (2012), which allows us to relax the well-known "Wallace neutrality" and make financial assets imperfect substitute so that sovereign bonds' purchases by the monetary authority have real effects in our model. In each EA region there are two types of households, "restricted" and "unrestricted". Restricted households can invest only in domestic long-term sovereign bond market and, because they are owner of domestic capital producers (joint with domestic unrestricted households), in domestic private physical capital. The purchase of long-term government bonds by the monetary authority reduces long-term interest rates and therefore induces restricted households to increase consumption and investment via the standard intertemporal substitution effect.

 $^{^{9}}$ For each region, size refers to the overall population, to the number of firms operating in each sector and, in the case of each EA region, the number of capital producers.

Unrestricted households (1) have access to the domestic short-term private bond and longterm sovereign bond markets, (2) trade a riskless private bond with RW households, and (3) invest in physical capital because they own domestic capital producers.

The latter accumulate private physical capital by demanding final investment goods subject to quadratic adjustment costs on investment change. They rent out capital to the domestic firms producing intermediate goods. They maximize profits with respect to capital and investment taking prices as given, and evaluate returns according to a weighted average of restricted and unrestricted households' stochastic discount factors (where the weights reflect the corresponding population shares). The (net) revenues are rebated in a lump-sum way to domestic restricted and unrestricted households according to their corresponding shares.

Households consume a final good which is a composite of intermediate non-tradable and tradable goods. The latter are domestically produced or imported. All households supply differentiated labor services to domestic firms and act as wage setters in monopolistically competitive labor markets by charging a mark-up over their marginal rate of substitution between consumption and leisure.

On the production side, there are perfectly competitive firms that produce two final goods (consumption and investment goods) and monopolistic firms that produce intermediate goods (firms are owned by domestic unrestricted households). The two final goods are sold domestically and are produced combining all available intermediate goods using a constant-elasticity-ofsubstitution (CES) production function. The two resulting bundles can have different compositions. Intermediate tradable and non-tradable goods are produced combining domestic public capital, private capital, and labor. The latter two production factors are assumed to be mobile across sectors. Intermediate tradable goods can be sold domestically and abroad. Since intermediate goods are differentiated, firms have market power and restrict output to create excess profits. We also assume that markets for tradable goods are segmented, so that firms can set a different price for each of the three markets. In line with other dynamic general equilibrium models of the EA (see, among the others, Christoffel *et al.* 2008 and Gomes *et al.* 2010), we include adjustment costs on real and nominal variables ensuring that consumption, production, and prices react in a gradual way to a shock. On the real side, habits and quadratic costs prolong the adjustment of consumption and investment, respectively. On the nominal side, quadratic costs make wages and prices sticky.¹⁰

2.2 Restricted households

There exists a continuum of restricted households j', with $j' \in (0, n\lambda_R]$, where $0 \leq \lambda_R \leq 1$. Their preferences are additively separable in consumption and labor effort. The generic restricted household j' receives utility from consumption $C_R(j')$ and disutility from labor $L_R(j')$. Following common practice in the New Keynesian literature, the assumption of cashless economy holds in the model. The household's expected lifetime utility is

$$E_0 \left\{ \sum_{t=0}^{\infty} \beta_R^t \left[\frac{\left(C_{R,t}(j') - h C_{R,t-1} \right)^{1-\sigma}}{(1-\sigma)} - \frac{L_{R,t}(j')^{1+\tau}}{1+\tau} \right] \right\},\tag{1}$$

where E_0 denotes the expectation conditional on information set at date 0, β_R is the discount factor ($0 < \beta_R < 1$), $1/\sigma$ is the elasticity of intertemporal substitution ($\sigma > 0$), and $1/\tau$ is the labor Frisch elasticity ($\tau > 0$). The parameter h (0 < h < 1) represents external habit formation in consumption.

Restricted households have access only to the market of long-term sovereign bonds. The budget constraint is

$$P_{t}^{L}B_{R,t}^{L}(j') - \sum_{s=1}^{\infty} \kappa^{s-1}B_{R,t-s}^{L}(j')$$

$$= \Pi_{t}^{prof}(j') + (1 - \tau_{t}^{\ell}) W_{R,t}(j') L_{R,t}(j')$$

$$-P_{t}(1 + \tau_{t}^{c}) C_{R,t}(j') - AC_{R,t}^{W}(j'),$$
(2)

where $B_{R,t}^{L}$ is the amount of long-term sovereign bonds, Π_{t}^{prof} is profit from ownership of the Home capital producers, $0 \leq \tau_{t}^{c} \leq 1$ is the tax rate on consumption. The long-term sovereign bonds have price $P_{L,t}$ and are formalized as perpetuities paying an exponentially decaying coupon $\kappa \in (0, 1]$, following Woodford (2001). Finally, households act as wage setters in a monopolistic competitive labor market. Each household j' supplies one particular type of labor services which is an imperfect substitute to services supplied by other households. It sets its nominal wage

 $^{^{10}}$ See Rotemberg (1982).

 $W_{R,t}$ taking into account of the labor income tax rate $0 \le \tau_t^{\ell} \le 1$, labor demand, and quadratic adjustment costs AC_R^W à la Rotemberg on the nominal wage $W_R(j')$:

$$AC_{R,t}^{W}(j') \equiv \frac{\kappa_{W}}{2} \left(\frac{W_{R,t}(j') / W_{R,t-1}(j')}{\Pi_{t-1}^{\alpha_{W}} \Pi_{EA}^{1-\alpha_{W}}} - 1 \right)^{2} W_{R,t} L_{R,t},$$
(3)

where $\kappa_W > 0$ and $0 \le \alpha_W \le 1$ are parameters regulating wage stickiness, the variable $\Pi_t \equiv P_{C,t}/P_{C,t-1}$ is the consumer price inflation rate, and $\bar{\Pi}_{EA}$ is the long-run inflation target of the EA monetary authority (assumed to be constant). The adjustment costs are proportional to the per-capita wage bill of restricted households, $W_{R,t}L_{R,t}$.¹¹

Restricted households are crucial for the PSPP to have real effects in our model. As they cannot arbitrage between short-term and long-term bonds, their consumption decisions depend only upon the long-term interest rate. Therefore, the monetary policy authority can affect their consumption and saving decisions by directly intervening in the secondary long-term sovereign bond market to change the long-term interest rate.

2.3 Unrestricted households

There exists a continuum of unrestricted households, indexed by j, with $j \in (n\lambda_R, n]$. These households have the same preferences as restricted households, thus they consume and supply labor. The only difference is the discount factor, β_U , which can be different from that of restricted households.

Home unrestricted households have access to multiple financial assets (all denominated in euro terms): the short-term (one-period) sovereign bond B^G , exchanged with the domestic government; the short-term private bond B^P , exchanged with REA unrestricted and RW households and paying the interest rate R^P ; the long-term sovereign bond B_U^L , exchanged with the domestic restricted households, domestic government and, because of the PSPP, the EA monetary authority. Thus, they have several opportunities to smooth consumption when facing a shock. The

 $^{^{11}}$ As the implied first order conditions are rather standard we do not report them to save on space. They are available upon request.

budget constraint of the generic unrestricted household j is

$$B_{t}^{G}(j) - B_{t-1}^{G}(j) R_{t-1}$$

$$+B_{t}^{P}(j) - B_{t-1}^{P}(j) R_{t-1}^{P}(1 - \phi_{t-1})$$

$$+P_{t}^{L} B_{U,t}^{L}(j) - \sum_{s=1}^{\infty} \kappa^{s-1} B_{U,t-s}^{L}(j) =$$

$$(1 - \tau_{t}^{\ell}) W_{U,t}(j) L_{U,t}(j) + \Pi_{t}^{P}(j) + \Pi_{t}^{prof}(j) - P_{t}(1 + \tau_{t}^{c}) C_{U,t}(j)$$

$$+TR_{t}(j) - AC_{U,t}^{W}(j) - AC_{U,t}^{B}(j),$$

$$(4)$$

where the short-term government bond B_t^G pays the EA monetary policy rate R_t . The dividends $\Pi_t^P(j)$ are from ownership of domestic monopolistic firms (claims to firms' profits are not internationally tradable). The term ϕ_t represents an exponential adjustment costs, needed to stabilize the position in the internationally traded bond.¹² The variable Π_t^{prof} is profit from ownership of the Home capital producers. The term TR_t represents lump-sum transfers from the government. Unrestricted households supply labor services under monopolistic competition and face quadratic adjustment costs $AC_{U,t}^W$ when setting nominal wages (the cost is similar to the one paid by restricted households, see eq. 3). They also pay adjustment costs $AC_{U,t}^B$ on long-term sovereign bond holdings.¹³

First order conditions imply no-arbitrage conditions for the unrestricted households.¹⁴ Thus, in equilibrium the interest rates paid by the different bonds are equal to the monetary policy rate R_t , net of the spreads induced by the longer maturity and the adjustment costs.¹⁵

$$\phi_{B,t} \equiv \phi_{b1} \frac{\exp\left(\phi_{b2}\left(B_t^P - \bar{B}^P\right)\right) - 1}{\exp\left(\phi_{b2}\left(B_t^P - \bar{B}^P\right)\right) + 1}, \text{ with } \phi_{b1}, \phi_{b2} > 0$$

 $^{13}\mathrm{We}$ assume a standard quadratic form for the adjustment cost, that is,

$$AC_{U,t}^{B}\left(j\right) \equiv \frac{\phi_{bL}}{2} \left(P_{t}^{L}B_{U,t}^{L}(j) - \bar{P}^{L}\bar{B}_{U}^{L}\right)^{2}, \text{ with } \phi_{bL} > 0,$$

 $^{^{12}}$ The adjustment cost is defined as

where B_t^P and \overline{B}^P are the period-by-period and steady-state positions of the representative Home unrestricted household, respectively. Both are taken as given in the maximization problem. A similar cost holds for the RW household.

where $\bar{P}^L \bar{B}_U^L$ is the (symmetric) steady-state value of the long-term sovereign bond. The adjustment cost guarantees that the bond holdings follow a stationary process and that the economy converges to the steady state.

 $^{^{14}}$ As the implied first order conditions are rather standard we do not report them to save on space. They are available upon request.

 $^{^{15}}$ See Chen *et al.* (2012) for the details. Our calibration implies that households can modify their financial positions without facing relevant adjustment costs.

2.4 Capital producers

There exists a continuum of mass $0 \le n \le 1$ of firms *e* that produce private physical capital. They optimally choose capital K_t and investment I_t to maximize profits subject to the law of capital accumulation, the adjustment costs on investment, and taking prices as given. The law of motion of capital accumulation is

$$K_t(e) = (1 - \delta) K_{t-1}(e) + (1 - AC_t^I(e)) I_t(e),$$

where $0 < \delta < 1$ is the depreciation rate. The adjustment cost on investment AC_t^I is

$$AC_{t}^{I}(e) \equiv \frac{\phi_{I}}{2} \left(\frac{I_{t}(e)}{I_{t-1}(e)} - 1 \right)^{2}, \text{ with } \phi_{I} > 0.$$

Capital producers rent existing physical capital stock $K_{t-1}(e)$ at the nominal rate R_t^K (on which they pay the tax rate $0 \le \tau_t^k \le 1$) to domestic firms producing intermediate tradable and nontradable goods. Investment is a final non-tradable good, composed of intermediate tradable (domestic and imported) and non-tradable intermediate goods. Capital producers buy it in the corresponding market at price P_I .¹⁶ Because of the adjustment costs on investment, a "Tobin's Q" holds.

When maximizing profits with respect to capital and investment, capital producers discount profits using the stochastic discount rates of restricted and unrestricted households, aggregated according to the corresponding population shares.

2.5 Public capital and firms' decisions

The production function of the generic firm i in the Home intermediate tradable sector is

$$Y_{T,t}(i) = K_{T,t}^{P}(i)^{\alpha_{1T}} L_{T,t}^{U}(i)^{\alpha_{2T}} L_{T,t}^{R}(i)^{\alpha_{3T}} (K_{G,t-1})^{1-\alpha_{1T}-\alpha_{2T}-\alpha_{3T}}$$

 $^{^{16}}$ As for the consumption basket, the investment bundle is a composite of tradable and non-tradable goods. The composition of consumption and investment goods can be different.

where $K_{T,t}^{P}(i)$ is private capital, which is supplied by the domestic capital producers, $L_{T,t}^{R}(i)$ and $L_{T,t}^{U}(i)$ represent labor supplied by, respectively, domestic restricted and unrestricted households, $K_{G,t-1}$ is public capital, accumulated by the domestic public sector, . The parameters $0 < \alpha_{iT} < 1$ $(i = 1, 2, 3), \alpha_{1T} + \alpha_{2T} + \alpha_{3T} < 1$, are the weights on private capital, unrestricted households' labor, and restricted households labor, respectively.

The firm optimally chooses demand for private capital and labor taking prices and the amount of public capital as given. Thus, firms do not demand public capital and there is no price or tariff paid for its use.

A similar production function holds for the generic firm i producing the intermediate nontradable good:

$$Y_{N,t}(i) = K_{N,t}^{P}(i)^{\alpha_{1N}} L_{N,t}^{U}(i)^{\alpha_{2N}} L_{N,t}^{R}(i)^{\alpha_{3N}} (K_{G,t-1})^{1-\alpha_{1N}-\alpha_{2N}-\alpha_{3N}}$$

For public capital projects, we follow Leeper *et al.* (2010) and in some simulations assume "time-to-build" (Kydland and Prescott, 1982): there is a delay between the authorization of a government spending plan and the completion of an investment project. The possibility of several periods of time-to-build in public capital implies that the government initiates investment projects that take N periods until they become productive and augment the public capital stock. Thus, the public capital is accumulated by the public sector according to

$$K_{G,t-1} = (1 - \delta_G) K_{G,t-2} + A_{I_G,t-1-N},$$

where $0 < \delta_G < 1$ is the depreciation rate, and $A_{I_G,t-1-N}$, with $N \ge 1$, is authorized government investment in period t-1-N. The time-to-build lags capture the idea that it takes time before a public investment is finished and, hence, can be effectively included in the public capital stock and affect the supply side of the economy. A "classic" example is the government that authorizes funding at time (quarter) t-8 for a highway that takes two years to build (N = 8). Then the highway cannot be considered as a part of the stock of public capital until quarter t ($K_{G,t-1}$ is used to produce goods in period t).

To capture the idea that spending outlays typically occur over time, we introduce the sequence

 $\{b_0, b_1, b_2, ..., b_{N-1}\}$ of the spending rates from the date the funding is authorized (date t - 8) to the period before project completion (date N - 1). For example, the highway may not be usable for two years but government investment increases during this time as construction of the highway takes place. Therefore, government investment actually implemented at time t is then given by

$$I_{G,t} = \sum_{n=0}^{N-1} b_n A_{I_G,t-n},$$
(5)

$$\sum_{n=0}^{N-1} b_n = 1. (6)$$

where the rate at which the construction takes place is parameterized by the *b*'s. In the case of a one-period time-to-build technology (as assumed for private investment), public investment outlaid in period *t* becomes productive in period t + 1, i.e. N = 1 and $I_{G,t} = A_{I_G,t}$.

2.6 Fiscal sector

Fiscal policy is set at the regional level. The government budget constraint is

$$B_{G,t}^{S} - B_{G,t-1}^{S}R_{t-1} + P_{L,t}B_{G,t}^{L} - \sum_{s=1}^{\infty} \kappa^{s-1}B_{G,t-s}^{L} \le P_{N,t}C_{G,t} + P_{I_G,t}I_{G,t} + TR_t - T_t,$$
(7)

where $B_{G,t}^S$, $B_{G,t}^L$ are short-term and long-term nominal sovereign bonds, respectively ($B_{G,t}^S$, $B_{G,t}^L > 0$ is public debt). The short-term bond is a one-period nominal bond issued in the domestic bond market that pays the (gross) monetary policy interest rate R_t . The implied gross yield to maturity at time t on the long-term bond is defined as

$$R_t^L = \frac{1}{P_t^L} + \kappa. \tag{8}$$

The variable $C_{G,t}$ represents government purchases of goods and services, $Tr_t > 0$ (< 0) are lump-sum transfers (lump-sum taxes) to households. Consistent with the empirical evidence, $C_{G,t}$ is fully biased towards the intermediate non-tradable good. Therefore, it is multiplied by the corresponding price index $P_{N,t}$.¹⁷ Given that the public investment has its own composition,

 $^{^{17}\}mathrm{See}$ Corsetti and Mueller (2006).

it is pre-multiplied by the public investment price deflator $P_{I_G,t}$. The investment in public capital $I_{G,t}$ is assumed, in line with empirical evidence, to have a composition that is more biased towards domestic goods. Thus, we assume that it has the same composition as the private consumption good. The same tax rates apply to every domestic household and capital producer (the latter pays the tax rate $0 \le \tau_t^k \le 1$ on return R_t^k on capital K_{t-1}). Total government revenues T_t from distortionary taxation are given by the identity

$$T_{t} \equiv \int_{0}^{n\lambda_{R}} \tau_{t}^{\ell} W_{t} \left(j'\right) L_{t} \left(j'\right) dj' \qquad (9)$$

$$+ \int_{n\lambda_{R}}^{n} \tau_{t}^{\ell} W_{t} \left(j\right) L_{t} \left(j\right) dj \qquad (9)$$

$$+ \int_{0}^{n} \tau_{t}^{k} R_{t}^{k} K_{t-1} \left(e\right) de \qquad (9)$$

$$+ \int_{0}^{n\lambda_{R}} \tau_{t}^{c} P_{t} C_{t} \left(j'\right) dj' \qquad (9)$$

The government follows a fiscal rule defined on lump-sum transfers to bring the short-term public debt as a percentage of domestic GDP, $b_G^s > 0$, in line with its long-run (steady-state) target \bar{b}_G^s .¹⁸ The rule is

$$\frac{TR_t}{TR_{t-1}} = \left(\frac{b_{G,t}^s}{\bar{b}_G^s}\right)^{-\phi_1},\tag{11}$$

where the parameter $-\phi_1$ is lower than zero ($\phi_1 > 0$), calling for a reduction (increase) in lump-sum transfer whenever the current-period short-term public debt (as a ratio to GDP) is above (below) the target. We choose lump-sum transfers to stabilize public finance as they are non-distortionary and, thus, allow a "clean" evaluation of the macroeconomic effects of public investment.

For long-term public debt, it is assumed for simplicity that its rate of change is the same as

$$GDP_t = P_t C_t + P_t^I I_t + P_t^{I_G} I_{G,t} + P_{N,t} C_{G,t} + P_t^{EXP} EXP_t - P_t^{IMP} IMP_t,$$
(10)

¹⁸The definition of nominal GDP is

where P_t , P_t^I , $P_t^{I_G}$, $P_{N,t}$, P_t^{EXP} , P_t^{IMP} are prices of private consumption, private investment, public investment, public consumption (given the assumption of fully biased composition towards intermediate non-tradable goods), exports, and imports, respectively.

that of the short-term public debt, so that the maturity composition of the overall public debt does not change.

Fiscal items other than (i) public deficit, (ii) public debt, (iii) lump-sum transfers, and (iv) public investment are kept at their corresponding initial (steady-state) levels when simulating the model.

2.7 Monetary authority

The EA monetary authority sets the (short-term) policy rate R_t according to a Taylor rule of the form

$$\left(\frac{R_t}{\bar{R}}\right)^4 = \left(\frac{R_{t-1}}{\bar{R}}\right)^{4\rho_R} \left(\frac{\Pi_{EA,t,t-3}}{\bar{\Pi}^4}\right)^{(1-\rho_R)\rho_\pi} \left(\frac{GDP_{EA,t}}{GDP_{EA,t-1}}\right)^{(1-\rho_R)\rho_{GDP}}.$$
 (12)

The parameter ρ_R ($0 < \rho_R < 1$) captures inertia in interest-rate setting, while the term \bar{R} represents the steady-state gross nominal policy rate. The parameters ρ_{π} and ρ_{GDP} are respectively the weights of yearly EA CPI inflation rate $\Pi_{EA,t,t-3} \equiv P_{C,t}/P_{C,t-4}$ (in deviation from the longrun steady-state target $\bar{\Pi}^4$) and the GDP gross growth rate ($GDP_{EA,t}/GDP_{EA,t-1}$). The CPI inflation rate is a geometric average of Home and REA inflation rates, with weights equal to the corresponding (steady-state) regional GDP (as a share of the steady-state EA GDP). EA GDP is the sum of Home and REA GDP. A similar equation describes monetary policy in the RW region.

In some simulations, the central bank resorts to non-standard measures, namely FG and QE. In the first case, the monetary authority credibly commits to keep the policy rate constant at its baseline (steady-state) level for a prolonged period of time and, thereafter, returns to follow the Taylor rule; in the second case, the central bank purchases long-term (domestic) sovereign bonds in the secondary markets to affect the long-term interest rates and hence the real side of the economy. The so called Wallace neutrality¹⁹ is relaxed by making short- and long-term sovereign bonds imperfect substitute in households' portfolios.²⁰ Specifically, we allow for preferred habitat investors: in each EA region some agents do not have access to short-term bonds and invest in

¹⁹See Wallace (1981).

 $^{^{20}}$ See for instance Chen *et al.* (2012)

physical capital and domestic long-term sovereign bonds: the monetary authority, by buying long-term sovereign bonds, lowers their yields and induces households to substitute investment in physical capital for sovereign bonds.²¹

The market clearing condition for the long-term sovereign bonds of the Home region is

$$\int_{0}^{n\lambda_{R}} B_{R,t}^{L}(j')dj' + \int_{n\lambda_{R}}^{n} B_{U,t}^{L}(j)dj + B_{PSPP,t}^{L} = B_{G,t}^{L},$$

where $B_{PSPP,t}^{L}$ are the central bank purchases. A similar condition holds for the REA region.

Finally, we also consider the case of the monetary authority permanently increasing its balance sheet through a QE that is directly proportional to the increase in public investment. We compare this policy measure with the distortionary tax-based financing of the public investment.

2.8 Equilibrium

In each country the initial asset positions, preferences, technologies, and budget constraints are the same for households belonging to the same type and firms belonging to the same sector. Moreover, profits from ownership of domestic monopolistically competitive firms are equally shared among unrestricted households. Profits from ownership of domestic capital producers are distributed to restricted and unrestricted households according to the corresponding population shares, and are equally shared within each type. Thus, we consider the representative household for each household type (restricted and unrestricted). Moreover, we consider the representative firm for each sector (final non-tradable, intermediate tradable, and intermediate non-tradable) and the representative capital producer. The implied symmetric equilibrium is a sequence of allocations and prices such that, given initial conditions and considered shocks, households and firms satisfy their corresponding first order conditions, the monetary rules, the fiscal rules, and the government budget constraints hold, and all markets clear.

 $^{^{21}}$ There is empirical evidence that supports the preferred habitat theory in the case of the EA (see Blattner and Joyce 2016 and Altavilla *et al.* 2015): the Eurosystem purchases create a scarcity in some bonds considered special by private investors, leading investors to push up prices and lower yields if these securities cannot easily be replaced with other securities featuring similar characteristics.

2.9 Calibration

The model is calibrated at quarterly frequency. We set some parameter values so that steadystate ratios are consistent with average EA national account data. For remaining parameters we resort to previous studies and estimates available in the literature.²²

Table 1 contains parameters for preferences and technology. Parameters with "*" and "**" are related to the REA and the RW, respectively. We assume perfect symmetry between the REA and the RW unless differently specified. The discount factor of EA unrestricted households is set to 0.9994, so that the steady-state short-term interest rate is equal to 0.25% on an annual basis. The discount factor of RW households is also set to 0.9994. The discount factor of restricted households determines the steady-state value of the long-term interest rate and is set to 0.995, so that in steady state the spread between short- and long-term bond is equal to 1.8pp. We set the intertemporal elasticity of substitution to 1, the Frisch elasticity of labor supply to 1/2, and the depreciation rate of private and public capital to 0.025, as customary in the literature. The habit motive for consumption implies a persistence of consumption patterns equal to 0.8. We set the elasticity of output to public capital to 0.1, in line with common practice (see Elekdag and Muir 2014 and De Jong *et al.* 2017) and with ample empirical evidence provided by Bom and Ligthart (2014).

In each EA region the share of restricted households is set to 0.25. Given the lack of microevidence on this share, we set it to get a response of investment to the (benchmark) PSPP around four times as large as the response of consumption, in line with standard business cycle facts, and at the same time to calibrate the adjustment cost on investment to a rather standard value (i.e., 7.50, as reported in Table 3), in line with Smets and Wouters (2003).²³

Tables 2, 3, and 4 report gross mark-up, adjustment costs, and parameters of the monetary and fiscal rules, respectively.

The parameter regulating the adjustment costs paid by the unrestricted household on deviations of long-term sovereign bond positions from steady-state levels, ϕ_{bL} , is set to 0.00047

 $^{^{22}}$ See the New Area Wide Model (NAWM, Christoffel *et al.* 2008) and Euro Area and Global Economy Model (EAGLE, Gomes *et al.* 2010), and the Global Economy Model (GEM, Pesenti 2008).

 $^{^{23}}$ We run robustness analysis. To save on space we do not include it in the paper. Specifically, when we reduce the share of restricted households to 0.15; the peak effect on EA GDP and inflation roughly halves. Qualitatively, results do not change.

and to 0.0008 in Home and REA, respectively. The parameters regulating the adjustment cost on private bond position, paid by Home unrestricted households and RW households, are set to 0.0015 and 0.003. These parameters have been calibrated following two criteria. First, they should not greatly affect the model dynamics and yet help to stabilize it. Second, the response of the interest rate on long-term sovereign bonds to the benchmark PSPP should be in line with existing evidence for the EA.²⁴

Table 4 reports the parametrization of the systematic feedback rules followed by the fiscal and monetary authorities. It is always lump-sum transfers to adjust to ensure stability of the debtto-GDP ratio. The corresponding parameter of the fiscal rule is set to 1.01. The central bank of the EA targets the contemporaneous EA-wide consumer price inflation (the corresponding parameter is set to 1.7) and the output growth (the parameter is set to 0.1). Interest rate is set in an inertial way and hence its previous-period value enters the rule with a weight equal to 0.92. The values are identical for the corresponding parameters of the Taylor rule in the RW.

Table 5 reports the great ratios, which are matched by the model steady state under our baseline calibration. We assume a zero steady-state net foreign asset position of each region. The sizes of Home and REA GDPs as shares of world GDP are set to 2.5% and to 19.1%, respectively. So the Home GDP is around 12% of EA GDP.

Short-term public debt (ratio to yearly GDP) is set to 13% for Home and 8% for the REA, respectively. Long-term public debt is set to 121% and 93% of (yearly) GDP for Home and the REA. Thus, total public debt as a share of GDP is 134% in Home and 101% in REA. We assume that in each country long-term sovereign bond holdings are equally shared between unrestricted and restricted households. The parameter κ is calibrated to match the average duration of the representative long-term EA sovereign bond, which is equal to 8 years.

Variables of the RW are set to values equal to those of corresponding REA variables.

The chosen calibration yields impulse response functions to a standard monetary policy shock (+0.25 basis points) for GDP and inflation in each EA region that are in line with the workhorse estimated models of the EA in the literature.²⁵

²⁴See Altavilla *et al.* (2015).

 $^{^{25}}$ See, for example, Gomes *et al.* (2010).

3 Simulated scenarios

In the first scenario the Home government increases public investment by 1% of the before-shock (steady-state) GDP for five years and, thereafter, gradually decreases it to the baseline level (the public investment follows an AR(1) process with a quarterly decay rate of 0.9). During the first five years, the fiscal rule (eq. 11) is not active; it start operating from the beginning of the sixth year, when lump-sum transfers to households endogenously adjust to stabilize the public debt-to-GDP ratio. The EA monetary policy rate follows the Taylor rule (12); alternatively, the policy rate remains constant for eight quarters at its baseline level and the Taylor rule becomes operative only thereafter (FG assumption). The comparison between the two alternative monetary policy responses allows to assess to what extent a more accommodative stance can increase the fiscal multiplier and mitigate the impact on the debt and the deficit ratios of the rise in government spending.

We compare the previous scenarios with those in which the increase (equal to 1% of the before-shock GDP) in public investment is simultaneously implemented in the Home and REA blocs; the assumptions on monetary policy are the same as above: in one case the short-term rate is determined by the Taylor rule; in another it is kept fixed for eight quarters; in the final case the stimulus coming from the low and fixed policy rate is reinforced by QE (the so-called Public Sector Purchase Programme, PSPP for short). Under the PSPP the central bank commits to purchase ≤ 180 billion per period for seven quarters (≤ 60 billion in the first quarter), as was the case for the PSPP announced in January 2015 (long-term sovereign bond purchases in Home and REA are proportional to the size of the corresponding region, measured as a share of the size of the EA).

We also analyze how the form of financing affects the impact on the economy of the increase in public investment by comparing the benchmark case of debt-financing with a hike in distortionary taxation. Moreover, we assess the role of the increase in sovereign spread when the stimulus is financed by issuing public debt.

Finally, we consider a case where the monetary authority buys and rolls over an amount of sovereign bonds equal to the increase in public investment. In this way, the monetary authority keeps low both the policy rates (with forward guidance) and the long-term interest rates (with the accommodative monetary stance).

We also perform a sensitivity analysis to assess the relationship between macroeconomic effectiveness and implementation efficiency of the investment programme, where the lack of efficiency is associated with low and slow accumulation of public investment and, thus, reduced impact on the supply side of the economy.

All simulations are run under perfect foresight, with households and firms considering fully credible the policy announcement.

4 Results

We initially describe the results for the case of a single-country increase in public investment. Subsequently, we report results for the case of a joint public investment boost in both EA regions.

4.1 Home-country increase in public investment

Figure 1 shows the quarterly responses of the main Home variables when only the domestic fiscal authority increases public investment (the REA public investment is constant at its baseline level). We consider both the case when the policy interest rate is set according to the Taylor rule (No FG) and the case when the interest rate is kept constant for the first eight quarters (FG). Economic activity in the Home country expands in both scenarios. Aggregate demand rises following the increase in public investment. Consumption is roughly constant, while private investment slightly decreases during the first five years, due to the increase in the long-term interest rate. Spending in infrastructures stimulates not only aggregate demand but also supply, via the accumulation of public capital, which enters the production function: a higher stock of public capital fosters labour productivity and boosts labor supply. The implied reduction in firms' current and expected marginal costs lowers the price of domestic goods and thus CPI inflation. The public sector deficit as a share of output increases, as the GDP gain does not generate tax revenues large enough to compensate for the higher expenditures. The deficit reduction starts from the sixth year, when the fiscal stimulus stops and the fiscal rule becomes active again. The public debt-to-annualized GDP ratio falls on impact. Thereafter, and crucially, rises and remains persistently above its baseline value, implying that the fiscal stimulus is not self-financing.²⁶ The euro slightly appreciates vis-à-vis the RW currency: the reason is that the EA monetary authority raises the policy rate more than the RW central bank does.²⁷

The expansionary effects are slightly larger in the FG scenario; the reason of so small a difference is that the standard Taylor rule in the No-FG scenario does not command a large increase in the policy rate in response to the fiscal stimulus, because the Home country is a relatively small part of the monetary union and its spillovers on inflation and economic activity in the REA are positive but contained (see Figure 2). Not surprisingly, the spillovers are slightly amplified by the accommodative monetary policy stance: the increase in REA inflation induces a decrease in the REA ex-ante real interest rate, which stimulates REA consumption and investment. The euro slightly depreciates vis-à-vis the RW currency, because the RW monetary authority raises the policy rate more than the EA authority. The deficit- and debt-to-GDP ratios are basically the same as those observed when monetary policy follows a Taylor rule.

Table 6, columns "Home increase" and "Home increase+FG", reports the first-, second-, and tenth-year (long run, LR) responses (they are averages of the corresponding quarterly responses). The single-country fiscal stimulus has long-lasting expansionary effects. Domestic consumption and investment benefit from the supply-side expansion and the associated increase in permanent income; exports rise because of price-competitiveness gains following the depreciation of the exchange in real terms. Imports increase as well, due to the increase in domestic demand.

To summarize, an increase in public investment occurring only in one country is expansionary. It basically has no effects on the rest of the monetary union if monetary policy responds to the improved business cycle conditions; it increases output and inflation slightly more at home and in the REA if instead the central bank keeps short-term interest rate constant.

 $^{^{26}}$ The ratio of public debt to GDP can fall on impact even though the deficit over GDP increases because of a "valuation effect" associated with the dynamics of the sovereign bond prices. The increase in long-term interest rate connected with the public investment programme leads to a decrease in the market price of sovereign bonds. Given that we model the long-term bonds as perpetuities, the value of the *overall* stock of public debt as a ratio to GDP decreases more than the increase in volume of the debt itself. As remarked also in the main text, this is an effect that benefits public debt only in the initial period of the simulation.

²⁷The nominal exchange rate dynamics is dictated by the uncovered interest parity condition, which links the EA and RW policy rates differential to the expected nominal exchange rate depreciation.

4.2 EA increase in public investment

Figure 3 shows the responses of the Home variables to an increase in public investment in both regions of the EA.

In a standard monetary policy regime (No FG), Home GDP benefits from the spillovers due to the fiscal boost in the REA and increases more than when infrastructure spending is raised only in the Home country. Exports increase, as they benefit from the expansion in REA consumption and investment (see Figure 4). The additional boost provided by foreign demand more than offsets the increase in supply and, thus, causes Home inflation to increase, unlike the case analysed in the previous subsection, where the rise in aggregate demand is not large enough to absorb the increase in supply. The public debt-to-GDP ratio initially decreases but then starts deteriorating and, in the long term (i.e., 10 years after the fiscal stimulus), it is higher than in the baseline.

Under FG the expansionary effects of the joint stimulus are further enhanced by the more accommodative monetary policy, as the nominal interest rate is kept constant at its baseline level for the first two years after the shock. For a given increase in inflation, the real interest rate decreases more than in the No-FG case, fueling a stronger expansion of consumption and (private) investment. Home Inflation steps up more as well, because of the large depreciation of the euro vis-à-vis the RW currency: the RW monetary authority, which faces expansionary spillovers from the EA, raises its policy rate more than the EA authority does, because the latter is committed to FG. The fiscal multiplier turns out to be slightly above 2 in both the short and in the long run.

The improvement in public finances is more pronounced and longer-lasting in the FG than in the No-FG scenario, because the fiscal stimulus is amplified by the accommodative stance of monetary policy and, thus, induces a larger increase in Home GDP. Ten years after the shock the public debt-to-GDP ratio is still lower than in the baseline, implying that the increase in public investment, in this case, is self-financing.

Figure 4 reports results for the REA variables. They are similar, qualitatively and quantitatively, to the results for the Home region. The effects of higher REA public spending on REA GDP are expansionary. The REA international relative prices (not reported) deteriorate, consistent with the excess aggregate supply that has to be absorbed by lowering the price of REA-produced goods.

Table 6 (columns "EA increase" and "EA increase+FG") contains the average values of the first, second, and tenth year (long run, LR) responses. Consistent with the results reported in the charts, the monetary stance does matter for the short-run macroeconomic effectiveness of the public investment increase. In the case of FG, the Home public debt-to-GDP ratio decreases in the short and, to some extent, in the long run. The corresponding ratio in the REA slightly increases in the long run. The ratio decreases in the Home region because Home tax rates are calibrated (and kept constant) to higher steady-state values than REA rates. This implies that, for a given increase in economic activity, fiscal revenues as a ratio to GDP increase to a larger extent in the Home region than in the REA.

Overall, we find that the EA-wide public investment increase has expansionary effects on the EA economy, and the Home region benefits from the expansionary spillovers associated with the increase in REA economic activity. The short-run expansionary effects are larger and the improvement in public finances is greater if the stance of monetary policy is accommodative.

4.3 EA-wide public investment increase and non-standard monetary policy measures: Beyond FG

In the model the non-standard monetary policy measures have real effects: the Wallace neutrality does not hold and short- and long-term bonds are imperfect substitutes. Thus, the monetary authority, by buying long-term sovereign bonds in the secondary market, is able to affect the long-term interest rate and to induce restricted households to substitute investment in physical capital for investment in sovereign bonds.

Figures 5 and 6 respectively show the macroeconomic effects on Home and REA regions of implementing the EA-wide stimulus when the EA monetary authority implements the twoyear FG on short-term interest rates (as in previous simulations) or, alternatively, the FG joint with the PSPP. The monetary authority buys long-term sovereign bonds and holds them up to maturity, after eight years. At the beginning of the ninth year, the monetary authority newly buys the same amount of bonds, so that its balance sheet does not change. Thereafter, the central bank holds the newly-bought bounds to maturity. The sovereign bond prices increase and consistently the long-term interest rates decrease (see eq. 8). Thus, households and firms have a larger incentive to substitute consumption and investment in physical capital for sovereign bond holdings. In both Home and REA regions the expansionary macroeconomic effects are larger than those under the assumption of implementing only the FG. The public deficit- and public debt-to-GDP ratios decrease favored by the large increase in GDP.

Table 7 reports the results for the first, second year, and tenth year (long run, LR) responses. The results are reported under standard monetary policy (No FG), FG only, and FG and PSPP. The PSPP strengthens the short- and long-run macroeconomic effectiveness of the public investment increase by persistently keeping the long-term interest rate below the baseline. The fiscal multiplier exceeds 3 and in the long run the public debt-to-GDP ratio turns out to be well below the baseline.

Overall, the PSPP reduces the long-term interest rates to a larger extent than the FG on short-term interest rates does, magnifying the macroeconomic impact of the increase in public investment. This is due to the fact that the expectations hypothesis, which makes short- and long-term bonds perfect substitutes (and, thus, non-standard monetary policy measures neutral) holds only for unrestricted households, that have access to both short- and long-term bonds. It does not hold for restricted households, that have access only to long-term sovereign bonds (preferred habitat assumption). The FG alone affects only the choices of unrestricted households, that are able to arbitrage away its impact on their portfolio choices. The PSPP instead affects also restricted households that do not have the same diversification opportunities as unrestricted households, thus magnifying the impact on the real variables. If there was no FG under the PSPP, its effectiveness would be great diminished due to residual arbitrage opportunities of unrestricted households.²⁸

4.4 EA-wide public investment financed with distortionary taxation

In the previous exercises it was assumed that the increase in public investment was initially financed by issuing public debt and, from the sixth year onwards, lump-sum taxes were allowed

 $^{^{28}}$ See Burlon *et al.* (2017) for an analysis of the effectiveness of the PSPP with different durations of the FG.

to rise to stabilise the debt-to-GDP ratio. To investigate how alternative funding sources bear upon the macroeconomic effectiveness of public investment spending, we look at the effects of the stimulus when it is paid by levying distortionary taxes. We consider an increase in distortionary taxation on consumption, labor and capital income. The FG assumption is maintained.

Distortionary tax rates are levied so as to obtain the same revenue - 1/3 of the increase in public investment (1% of initial GDP) - from each source: consumption, capital and labour income. The Home (REA) tax rates rise from $\tau^l = 0.426$ ($\tau^{l*} = 0.346$) to $\tau^l_t = 0.432$ ($\tau^{l*}_t = 0.352$), from $\tau^k = 0.349$ ($\tau^{k*} = 0.259$) to $\tau^k_t = 0.363$ ($\tau^{k*}_t = 0.273$), and from $\tau^c = 0.168$ ($\tau^{c*} = 0.203$) to $\tau^c_t = 0.173$ ($\tau^{c*}_t = 0.209$). The tax rates exogenously rise for five years, i.e., for the whole duration of the increase in public investment; thereafter lump-sum taxes are used to control debt developments.

Table 8 reports the simulation results. Compared with the benchmark scenario (column "EA increase+FG"), financing the increase in public investment through higher distortionary taxes leads to milder expansionary and inflationary effects in the first years. Higher distortionary taxation has negative income and substitution effects, inducing, *ceteris paribus*, households to reduce consumption, decrease the supply of labor, and invest less in physical capital. Thus, distortionary taxes attenuate the expansionary effects of the increase in productive capacity associated with a larger stock of public capital. With a notable exception, in the longer term the differences with respect to the benchmark scenario vanish, as the hike in distortionary taxation is temporary. The exception is represented by public finances: higher fiscal revenues decrease public deficit in the initial years, reducing public debt in the long run by more than in the benchmark case.

4.5 Increase in sovereign spread

The previous section suggests that in the short run the stimulus is more expansionary if financed by issuing debt. However, if the fiscal authority cannot credibly commit to the stabilization of public finances or if non-fundamental shocks can shift the demand for sovereign bonds, the sovereign risk premia may well rise in response to the announcement of a large program of fiscal stimulus, especially in countries with a high initial level of public debt. Considerably higher long-term interest rates connected with high risk premia can easily undo the expansionary effect of a debt-financed fiscal stimulus, worsening the future level of public deficits through lower fiscal revenues and potentially triggering a further deterioration of sovereign risk premia.

In this section we evaluate how the macroeconomic effectiveness of increasing Home public investment is affected by an increase in Home sovereign spread. We assume that the higher public investment is financed by issuing new debt and that the Home sovereign default risk affects Home consumption and investment decisions by creating a wedge between the risk-free rate and the government bonds yield. The spreads on both policy and long-term interest rates is assumed to increase by around 0.75 (annualized) bp during the first five years, in line with the empirical regularities between sovereign spread and public debt as documented in Corsetti et al. (2012).²⁹

Figure 7 reports the results. We assume that there is no FG and, thus, that the Taylor rule always holds. The case of spread increase (labeled "No FG & spread") is compared with the one in which the spread does not increase (labeled "No FG"). The latter is the same scenario as the one reported in Figure 1. The increase in Home spread and, thus, Home interest rates counterbalances the stimulating effect of the increase in Home public investment. Higher interest rates induce Home households and firms to reduce consumption and investment. Home GDP increases to a lower extent in the short run. Public deficit and public debt increase to a larger extent, because of the increase in interest spending and the lower increase in GDP (which reduces the increase in tax revenues).

Overall, the simulation suggests that public investment becomes a viable tool to stimulate the economy only when its increase is part of a plan that guarantees the stability of public finances and the sustainability of public debt.

 $^{^{29}}$ We do not consider here the ex-post consequences of an actual default. As in Corsetti et al. (2012), the model does not allow for a strategic default, that would result from an explicit decision of the policymaker, comparing costs and benefits of the default. Thus, the premium is not microfounded. This is a deliberate choice, to make the model tractable. The link between sovereign risk premium and the expected path of public debt is imposed, (1) building on the result (from the literature on strategic default) that in equilibrium the probability of default increases in the level of debt, and (2) implicitly assuming that there are limits to credible commitment on the part of fiscal policymakers. Equivalently, as in Corsetti et al. (2012), we assume that the government can make use of non-distortionary taxation (lump-sum transfers) so that, in case of a sovereign default, government bond holders would be compensated for their loss. Hence, while actual ex-post default is neutral, the ex-ante probability of default is crucial for the pricing of government debt and has real effects.

4.6 EA-wide public investment under monetary accommodation

We consider now a case where the monetary authority keeps low both the policy rates (with forward guidance) and the long-term interest rates. Different from the PSPP, the expansion of the balance sheet of the central bank is proportional to cost of the investment plan and is permanent. The monetary authority buys sovereign bonds to the extent that they are issued to finance the public investment and rolls them over to expand its balance sheet. The interest rates thus remain at a relatively low level despite the onset of the public investment program.

The last column of Table 8 reports the simulation results. Compared with the benchmark scenario (column "EA increase+FG"), an increase in public investment associated with an accommodative monetary policy stance leads to a more expansionary stimulus as it avoids the increase in interest rates both in the short end and in the long end of the yield curve. Its effects are also larger than those of the benchmark scenario, where the issuance of sovereign bonds partially crowds out consumption and investment because of the step-up in long-term interest rates. In the case of the monetary accommodation, there is no increase in long-term interest rates, which on the contrary slightly decrease as households invest part of their additional savings in sovereign bonds. The public debt-to-GDP ratio decreases substantially due to the increase in the GDP, and, in the short and medium run, slightly more than in the case of distortionary taxation. This is due to the fact that the initial increase in distortionary taxes reduces the GDP increase in the short and medium run, because households and firms are induced to postpone consumption, investment, and labor demand, to periods in which the tax rates return to their baseline levels. Consistent with that, the increase in GDP is less front-loaded in the case of increasing taxes. Thus, the long-run decrease in the public debt-to-GDP ratio is slightly larger when tax rates are increased.

Overall, results suggest that coordination between fiscal and monetary authority, as when the budget expansion is associated with an accommodative monetary policy stance, enhances the GDP response to the stimulus.

5 Sensitivity analysis

The results presented in the previous section rest on the assumptions that public capital enters the production function and the investment programme is implemented efficiently. We now evaluate the effects of a EA-wide increase in public investment and PSPP implementation when: i) the increase in public investment affects aggregate demand but has no impact on aggregate supply (i.e., we do not allow public investment expenditures to contribute to public capital accumulation); ii) investment increases the stock of public capital only with a lag (time-tobuild), i.e., we assume that it takes five years for the new capital goods to be delivered and become fully productive (in terms of eq. (5) it implies that in every quarter, a fraction equal to 5% of the authorized outlays does actually occur; thus, parameters $\{b_i\}_{i=1}^{N-1}$ are set to 1/20, and N is set to 20).³⁰ In both scenarios, the EA short-term interest rate is kept constant at the baseline level during the first eight quarters.

Table 9 reports the simulation results. The effects on GDP and the aggregate demand components are lower than in the benchmark scenario. As public spending does not affect capital accumulation, the long-run impact is negligible and entirely dependent on the response of households' labor supply; the short-run effects are modest as well: households and firms anticipate a subdued increase in permanent income and accordingly consume less. Similarly, in the time-to-build case, the GDP gain is lower in the short run, as it takes time to accumulate public capital. In both scenarios, public debt in the home country decreases less in both the short and the long run, as a consequence of the lower increase in GDP.

The results presented in Table 9 confirm that the benefits accruing from increasing government capital expenditure depend on (i) what goods are actually purchased (i.e. productive or unproductive goods) and on (ii) whether the investment plan is implemented efficiently (i.e. with no undue delays).

 $^{^{30}}$ In this way we capture implementation delays associated with the administrative procedures needed for planning, bidding, and contracting.

6 Conclusions

We have evaluated the macroeconomic impact of increasing spending in infrastructures in the EA under alternative assumptions about its financing.

Results suggests that the EA-wide fiscal stimulus can be rather effective in supporting economic activity in both the short and the long run. First, EA-wide stimuli are more effective than unilateral (region-specific) stimuli. Second, under EA-wide stimulus, the fiscal multiplier is close to 2 if the forward guidance (FG) on the short-term policy rate holds. Third, if the monetary authority keeps lower both the policy rates (with FG) and the long-term interest rates (with quantitative easing), the fiscal multiplier exceeds 3 at peak and investment spending is self-financing. Fourth, the financing method is relevant. Debt financing, in particular under an accommodative monetary stance and if the sovereign spreads do not increase, is more growthfriendly than tax financing in the short run (but not in the long run). Fifth, the effectiveness of the fiscal stimulus is larger if government spending is directed towards productive goods and its implementation occurs efficiently and without delays.

Our paper supports the view that spending on public capital can promote growth and provide benefits to society. It is therefore consistent with a large part of the recent literature, which includes the analysis presented in International Monetary Fund (2014a). There are however studies that reach opposite conclusions. Garin (2016) studies the impact of the American Recovery and Reinvestment Act's highway spending on county-level employment and finds basically zero effect four years after the start of the recession: some jobs were created in the counties of the contractors but not in those where the highways were built. The Japanese experience of the 1990s may also be considered as a cautionary tale: it certainly produced engineering marvels, but no material contribution to the economic recovery. The long-run beneficial impact of public capital accumulation is far from certain either. Warner (2014) examines whether big infrastructure and public capital spending programmes have succeeded in accelerating economic growth in low-income countries. The evidence he collects shows only a weak positive association between investment spending and growth, with no long-term impact. Pritchett (2000) argues that vast anecdotal experience suggests that many governments' investment efforts are much less productive than private ones. The available evidence is also not easy to interpret, due to the presence of reverse causality: not only might public investment stimulate growth, higher growth may also lead to higher demand for infrastructures. Moreover, the macroeconomic impact is dependent on conditions related to its implementation (e.g., the quantity and quality of the capital stock already in place; whether spending aims at alleviating bottlenecks; whether the project are politically viable at the local level), which explains why the effects of public investment differ across countries, regions, and sectors. It is therefore not surprising that the literature contains a relatively wide range of conflicting estimates, as described for instance by the review conducted by Sturm *et al.* (1998), who note that there are studies claiming that the marginal product of public capital is much higher than that of private capital, others suggesting that it is roughly equal, and still others finding that it is well below that of private capital and, in some cases, even negative.

The evidence presented in the paper is mostly based on the response of economic activity to the fiscal stimulus. No attempt is made to assess its welfare implications. The analysis of the welfare implications is left for future research.
References

- Abiad, A., D. Furceri, and P. Topalova (2016). The macroeconomic effects of public investment: Evidence from advanced economies. *Journal of Macroeconomics*, vol. 50, pages 224–240.
- [2] Altavilla, C., G. Carboni, and R. Motto (2015). Asset purchase programmes and financial markets: lessons from the euro area. ECB Working Paper no. 1864, November.
- [3] Banca d'Italia (2012). The efficiency of infrastructure spending. Workshop and conferences series, no. 10 (only in Italian).
- [4] Blanchard, O., C. Erceg, and J. Lindé (2014). The Euro Area Recovery: Should the Core Expand Spending to Help the Periphery?, mimeo.
- [5] Blattner, T., and M. Joyce (2016). Net debt supply shocks in the euro area and the implications for QE. ECB Working Paper no. 1957, September.
- [6] Bom, P. R. D., J. E. Ligthart (2014). What have we learned from three decades of research on the productivity of public capital? *Journal of Economic Surveys*, vol. 28(5), pages 889–916.
- [7] Bouakez, H., M. Guillard, and J. Roulleau-Pasdeloup (2017). Public investment, time to build, and the zero lower bound. *Review of Economic Dynamics*, vol. 23, pages 60-79.
- [8] Burlon, L., A. Gerali, A. Notarpietro, and M. Pisani (2017). Macroeconomic effectiveness of non-standard monetary policy and early exit. A model-based evaluation. *International Finance*, vol. 20 (2), pages 155173.
- [9] Chen, H., V. Curdia, and A. Ferrero (2012). The macroeconomic effects of large-scale asset purchase programmes. *The Economic Journal*, vol. 122, pages 289–315.
- [10] Christoffel, K., G. Coenen, and A. Warne (2008). The New Area-Wide Model of the Euro Area: A Micro-Founded Open-Economy Model for Forecasting and Policy Analysis. ECB Working Paper 944, European Central Bank.

- [11] Coenen, G., R. Straub, and M. Trabandt (2013). Gauging the effects of fiscal stimulus packages in the euro area. *Journal of Economic Dynamics and Control*, vol. 37, pages 367– 382.
- [12] Corsetti, G., and G. J. Mueller (2006). Twin Deficits: Squaring Theory, Evidence and Common Sense. *Economic Policy*, vol. 48, pages 597–638, October.
- [13] Corsetti, G., K. Kuester, A. Meier, and G. Mueller (2012). Sovereign risk, fiscal policy and macroeconomic stability. IMF WP 12/33.
- [14] De Jong, J., M. Ferdinandusse, J. Funda, and I. Vetlov (2017). The effect of public investment in Europe: a model-based assessment. ECB Working Paper 2021, European Central Bank.
- [15] Draghi, M. (2014). Keynote speech at the Eurofi Financial Forum. Milan, 11 September 2014.
 Available at: https://www.ecb.europa.eu/press/key/date/2014/html/sp140911_1.
 en.html.
- [16] Elekdag, S., and D. Muir (2014). Das Public Kapital: How Much Would Higher German Public Investment Help Germany and the Euro Area? IMF Working Paper 14/227, International Monetary Fund.
- [17] European Commission (2015). The Five Presidents' Report: Completing Europe's Economic and Monetary Union. Available at: https://ec.europa.eu/priorities/publications/ five-presidents-report-completing-europes-economic-and-monetary-union_en.
- [18] Garin, A. (2016). Putting America to Work, Where? The Limits of Infrastructure Construction as a Locally-Targeted Employment Policy. Taubman Center Working Paper WP 2016/01
- [19] Gomes, S., P. Jacquinot, and M. Pisani (2010). The EAGLE. A Model for Policy Analysis of Macroeconomic Interdependence in the Euro Area. ECB Working Paper 1195, European Central Bank.
- [20] In 't Veld, J. (2016). Public investment stimulus in surplus countries and their euro area spillovers. ECFIN European Economy Economic Brief 016, August.

- [21] International Monetary Fund (2014a). World Economic Outlook, Chapter 3, April.
- [22] International Monetary Fund (2014b). World Economic Outlook, Chapter 3, October.
- [23] International Monetary Fund (2015). Making public investment more efficient. IMF Policy Papers, June.
- [24] Kydland, F.E., and E.C. Prescott (1982). Time to build and aggregate fluctuations. *Econo*metrica 50, pages 1345–1370.
- [25] Leeper, E., T.B. Walker, and S.C.S. Yang (2010). Government Investment and Fiscal Stimulus. *Journal of Monetary Economics*, Vol. 57, pages 1000–1012.
- [26] OECD (2016). OECD Economic Outlook, no. 99, June.
- [27] Pesenti, P. (2008). The Global Economy Model (GEM): Theoretical Framework. IMF Staff Papers, vol. 55(2), pages 243–284.
- [28] Pritchett, L. (2000). The tyranny of concepts: CUDIE (Cumulated, Depreciated, Investment Effort) is not capital. *Journal of Economic Growth*, vol. 5, pages 361–384.
- [29] Rotemberg, Julio J. (1982). Monopolistic Price Adjustment and Aggregate Output. Review of Economic Studies, vol. 49, pages 517–531.
- [30] Smets, F., and R. Wouters (2003). An Estimated Dynamic Stochastic General Equilibrium Model of the Euro Area. *Journal of the European Economic Association*, MIT Press, vol. 1(5), pages 1123–1175.
- [31] Sturm JE., J. De Haan, G.H. Kuper (1998). Modelling Government Investment and Economic Growth: A Review and Some New Evidence. In: Shibata H., Ihori T. (eds) The Welfare State, Public Investment, and Growth. Springer, Tokyo.
- [32] Visco, I. (2014). World Savings Day Address by the Governor of the Bank of Italy Ignazio Visco- Rome - ACRI - Association of Italian Savings Banks. 31 October 2014. Available at: https://www.bancaditalia.it/pubblicazioni/interventi-governatore/ integov2014/visco-31102014-eng.pdf?language_id=1.

- [33] Wallace, N. (1981). A Modigliani Miller Theorem for Open Market Operations. The American Economic Review, vol. 71(3), pages 267–274.
- [34] Warner, A. M. (2014). Public Investment as an Engine of Growth. IMF Working Paper 14/148, International Monetary Fund, Washington.
- [35] Woodford, M. (2001). Fiscal requirements for price stability. Journal of Money, Credit and Banking, vol. 33, pages 669–728.

Parameter	Н	REA	RW
Discount factor $\beta_U, \beta_U^*, \beta^{**}$	0.9994	0.9994	0.9994
Discount factor β_R, β_R^*	0.995	0.995	_
Intertemporal elasticity of substitution $1/\sigma$	1.0	1.0	1.0
Share of restricted households λ_R	0.25	0.25	_
Inverse of Frisch Elasticity of Labor Supply τ	2.0	2.0	2.0
Habit h	0.8	0.8	0.8
Depreciation rate of (private and public) capital δ	0.025	0.025	0.025
Tradable Intermediate Goods			
Substitution between factors of production $\xi_T, \xi_T^*, \xi_T^{**}$	0.90	0.90	0.90
Share of private capital $\alpha_{1T}, \alpha_{1T}^{**}, \alpha_{1T}^{**}$	0.3	0.3	0.3
Share of unrestricted HH's labor $\alpha_{2T}, \alpha_{2T}^*, \alpha_{2T}^{**}$	0.5	0.5	0.7
Share of restricted HH's labor $\alpha_{3T}, \alpha_{3T}^*$	0.1	0.1	_
Share of public capital $1 - \alpha_{1T} - \alpha_{2T} - \alpha_{3T}, 1 - \alpha_{1T}^* - \alpha_{2T}^* - \alpha_{3T}^*$	0.1	0.1	_
Non tradable Intermediate Goods			
Substitution between factors of production $\xi_N, \xi_N^*, \xi_N^{**}$	0.90	0.90	0.90
Share of private capital $\alpha_{1N}, \alpha_{1N}^*, \alpha_{1N}^{**}$	0.3	0.3	0.3
Share of unrestricted HH's labor $\alpha_{2N}, \alpha_{2N}^*, \alpha_{2N}^{**}$	0.5	0.5	0.7
Share of restricted HH's labor $\alpha_{3N}, \alpha_{3N}^*$	0.1	0.1	_
Share of public capital $1 - \alpha_{1N} - \alpha_{2N} - \alpha_{3N}, 1 - \alpha_{1N}^* - \alpha_{2N}^* - \alpha_{3N}^*$	0.1	0.1	_
Final consumption goods			
Subst. btw. dom. and imported goods $\phi_A, \phi_A^*, \phi_A^{**}$	1.50	1.50	1.50
Bias towards domestic tradable goods a_H, a_F^*, a_G^{**}	0.68	0.59	0.90
Subst. btw. tradables and nontradables $\rho_A, \rho_A^*, \rho_A^{**}$	0.50	0.50	0.50
Bias towards tradable goods a_T, a_T^*, a_T^{**}	0.68	0.50	0.50
Final private investment goods			
Subst. btw. dom. and imported goods $\phi_E, \phi_E^*, \phi_E^{**}$	1.50	1.50	1.50
Bias towards domestic tradable goods v_H, v_F^*, v_G^{**}	0.50	0.49	0.90
Subst. btw. tradables and nontradables $\rho_E, \rho_E^*, \rho_E^{**}$	0.50	0.50	0.50
Bias towards tradable goods v_T, v_T^*, v_T^{**}	0.78	0.70	0.70
Final public investment goods			
Subst. btw. dom. and imported goods ϕ_A, ϕ_A^*	1.50	1.50	_
Bias towards domestic tradable goods a_H, a_F^*	0.68	0.59	_
Subst. btw. tradables and nontradables ρ_A, ρ_A^*	0.50	0.50	_
Bias towards tradable goods a_T, a_T^*	0.68	0.50	_
Tax rates			
Labor tax τ^l, τ^{l*}	0.426	0.346	0.346
Capital tax τ^k, τ^{k*}	0.349	0.259	0.259
Consumption tax τ^c, τ^{c*}	0.168	0.203	0.203

Note: H=Home;REA=rest of the euro area; RW= rest of the world. "*" refers to REA, "**" to RW

Table 2: Gross Mark-ups.

Mark-ups and Elasticities of Substitution									
	Tradables	nontradables	Wages						
Η	$1.08 \ (\theta_T = 13.32)$	$1.29 \ (\theta_N = 4.44)$	1.60 ($\psi = 2.65$)						
REA	1.11 $(\theta_T^* = 10.15)$	$1.24 \ (\theta_N^* = 5.19)$	1.33 $(\psi^* = 4)$						
RW	1.11 $(\theta_T^{**} = 10.15)$	$1.24 \ (\theta_N^{**} = 5.19)$	1.33 $(\psi^{**} = 4)$						
Note: I	H=Home; REA=rest of t	he euro area; RW= res	st of the world. "*" refers						

to REA, "**" to RW

Table 5. Real and Nommal Aujustment Costs.										
Parameter	Н	REA	RW							
Real Adjustment Costs										
Investment ϕ_I , ϕ_I^*, ϕ_I^{**}	7.50	7.50	7.50							
Adjustment Costs on bonds										
Households' long-term bond positions ϕ_{b^L} , $\phi_{b^L}^*$	0.00047	0.0008	_							
Households' short-term private bond positions										
$\phi_{b1}, \ \phi_{b1}^{**}$	0.0015	_	0.0015							
ϕ_{b2},ϕ_{b2}^{**}	0.003	_	0.003							
Nominal Adjustment Costs										
Wages $\kappa_W, \kappa_W^*, \kappa_W^{**}$	400	400	400							
Home produced tradables $\kappa_H, k_H^*, k_H^{**}$	400	400	50							
REA produced tradables κ_H , k_H^* k_H^{**}	400	400	50							
RW produced tradables κ_H , k_H^* , k_H^*	50	50	400							
Non-tradables κ_N , κ_N^* , κ_N^{**}	600	600	600							

Table 3: Real and Nominal Adjustment Costs.

Note: H=Home; REA=rest of the euro area; RW= rest of the world. "*" refers to REA, "**" to RW

Table 4: Fiscal and Monetary Policy Rules.

Parameter	Η	REA	EA	RW
Fiscal policy rule				
$\phi_1, \phi_1^*, \phi_1^{**}$	1.01	1.01	-	1.01
Common monetary policy rule	-	-		
Lagged interest rate ρ_R, ρ_R^{**}	-	-	0.92	0.92
Inflation $\rho_{\Pi}, \rho_{\Pi}^{**}$	-	-	1.70	1.70
GDP growth $\rho_{GDP}, \rho_{GDP}^{**}$	-	-	0.10	0.10

Note: H=Home; REA=rest of the euro area; RW= rest of the world. "* " refers to REA, "** " to RW

	Η	REA	RW
Macroeconomic variables			
Private consumption	63.3	60.7	60.2
Public consumption	20.0	20.0	20.0
Private investment	14.1	16.8	17.2
Public investment	3.0	3.0	0.0
Imports	27.2	19.1	4.8
Net Foreign Asset Position	0.0	0.0	0.0
GDP (share of world GDP)	2.5	19.1	78.4
Short-term public debt (ratio to annual GDP)	13.2	8.0	_
Long-term public debt (ratio to annual GDP)	121.3	93.3	_

Table 5: Main macroeconomic variables (ratio to GDP).

Note: H=Home; REA=rest of the euro area; RW= rest of the world.

Table 6: Public investment increase and FG.

	Ho	me incre	ease	Home	increas	e+FG	E	A increa	se	EA	increase	+FG
	1st	2nd	LR	1st	2nd	LR	1st	2nd	LR	1st	2nd	LR
Home												
GDP	0.68	0.80	1.54	0.73	0.85	1.54	0.94	1.04	1.70	1.94	2.20	1.84
Inflation	-0.07	-0.16	0.04	-0.03	-0.13	0.04	0.19	0.09	-0.04	1.13	0.79	-0.04
Short-term interest rate	0.01	0.01	0.00	0.00	0.00	0.00	0.11	0.18	-0.09	0.00	0.00	-0.09
Long-term interest rate	0.13	0.15	0.07	0.13	0.14	0.06	0.14	0.14	-0.02	0.02	0.03	-0.06
Consumption	-0.07	-0.06	1.00	-0.03	-0.01	1.01	0.12	0.20	1.38	0.97	1.28	1.53
Investment	-0.55	-1.06	1.76	-0.44	-0.90	1.75	-0.28	-0.63	2.63	1.90	2.54	2.47
Export	-0.02	0.35	1.97	0.03	0.39	1.98	0.46	0.66	2.21	1.59	1.52	2.43
Import	1.17	0.90	0.31	1.21	0.95	0.31	1.32	1.20	1.16	2.09	2.31	1.15
Labor	0.69	0.35	0.15	0.77	0.43	0.15	1.12	0.74	0.17	2.75	2.47	0.16
Real wage	0.14	0.35	1.03	0.14	0.36	1.03	0.15	0.41	1.33	0.10	0.46	1.45
Real exch. rate Home-REA	0.01	0.03	0.18	0.01	0.04	0.18	0.00	0.01	0.04	0.02	0.05	0.05
Real exch. rate Home-RW	0.01	0.05	0.31	0.04	0.07	0.31	0.11	0.11	0.39	0.70	0.38	0.44
Public deficit	1.00	1.08	-0.44	0.97	1.05	-0.43	0.89	0.95	-0.50	0.34	0.28	-0.25
Public debt	-1.53	-0.49	3.80	-1.59	-0.64	3.63	-2.17	-1.47	2.38	-3.43	-4.59	-0.89
REA												
GDP	0.01	-0.00	0.01	0.06	0.05	0.01	0.86	0.91	1.52	1.84	1.93	1.62
Inflation	0.00	0.00	0.00	0.05	0.04	0.00	0.24	0.13	-0.05	1.32	0.87	-0.05
Long-term interest rate	0.00	0.00	-0.00	0.00	0.00	-0.00	0.14	0.15	-0.02	0.04	0.05	-0.05
Real exch. rate REA-RW	-0.03	-0.01	0.02	0.11	0.04	0.03	0.53	0.47	1.68	3.43	1.54	1.84
Public deficit	0.01	0.01	-0.00	-0.01	-0.00	0.00	0.92	0.98	-0.49	0.44	0.41	-0.29
Public debt	-0.04	-0.02	-0.04	-0.10	-0.14	-0.10	-1.51	-0.72	2.78	-2.63	-3.33	0.37

Note: Home increase is 1% (of Home steady-state GDP) increase in the Home public investment. EA increase is a 1% (of steady-state EA GDP) increase in EA-wide public investment. FG represents a EA monetary policy rate kept at its baseline level during the first two years. Average values are reported. 1st= 1st year; 2nd= 2nd year; LR=long run (10 year). REA=rest of the EA. All variables are % dev. from steady state; inflation, interest rate, public deficit/GDP and public debt/(4*GDP) as % point dev. from steady state. For real exchange rates, +=depreciation.

	EA increase			EA	increase	+FG	FG and PSPP		
	1st	2nd	LR	1st	2nd	LR	1st	2nd	LR
Home									
GDP	0.94	1.04	1.70	1.94	2.20	1.84	3.03	3.55	2.63
Inflation	0.19	0.09	-0.04	1.13	0.79	-0.04	2.14	1.65	-0.06
Short-term interest rate	0.11	0.18	-0.09	0.00	0.00	-0.09	0.00	0.00	-0.04
Long-term interest rate	0.14	0.14	-0.02	0.02	0.03	-0.06	-0.45	-0.45	-0.34
Consumption	0.12	0.20	1.38	0.97	1.28	1.53	1.96	2.59	2.09
Investment	-0.28	-0.63	2.63	1.90	2.54	2.47	5.40	8.63	4.45
Export	0.46	0.66	2.21	1.59	1.52	2.43	2.53	2.11	3.46
Import	1.32	1.20	1.16	2.09	2.31	1.15	3.40	4.43	1.72
Labor	1.12	0.74	0.17	2.75	2.47	0.16	4.56	4.50	0.33
Real wage	0.15	0.41	1.33	0.10	0.46	1.45	0.09	0.57	1.99
Real exch. rate Home-REA	0.00	0.01	0.04	0.02	0.05	0.05	0.04	0.09	0.06
Real exch. rate Home-RW	0.11	0.11	0.39	0.70	0.38	0.44	1.17	0.53	0.64
Public deficit	0.89	0.95	-0.50	0.34	0.28	-0.25	-0.61	-0.83	-0.08
Public debt	-2.17	-1.47	2.38	-3.43	-4.59	-0.89	-1.67	-5.32	-6.68
REA									
GDP	0.86	0.91	1.52	1.84	1.93	1.62	2.92	3.15	2.34
Inflation	0.24	0.13	-0.05	1.32	0.87	-0.05	2.49	1.80	-0.07
Long-term interest rate	0.14	0.15	-0.02	0.04	0.05	-0.05	-0.42	-0.41	-0.32
Real exch. rate REA-RW	0.53	0.47	1.68	3.43	1.54	1.84	5.68	1.96	2.71
Public deficit	0.92	0.98	-0.49	0.44	0.41	-0.29	-0.37	-0.51	-0.14
Public debt	-1.51	-0.72	2.78	-2.63	-3.33	0.37	-1.45	-4.20	-4.11

Table 7: EA public investment, FG, and PSPP.

Note: EA public investment increase is a 1% (of EA steady-state GDP) increase in EA-wide public investment. FG represents a EA monetary policy rate kept at its baseline level during the first two years. PSPP is the implementation of the Public Sector Purchase Programme. Average values are reported. 1st= 1st year; 2nd= 2nd year; LR=long run (10 year). REA=rest of the EA. All variables as % dev. from steady state; inflation, interest rate, public deficit/GDP and public debt/(4*GDP) as % point dev. from steady state. For real exchange rates, +=depreciation.

	EA increase+FG			Ta	ax increa	ase	Monetary accommodation		
	1st	2nd	LR	1st	2nd	LR	1st	2nd	LR
Home									
GDP	1.94	2.20	1.84	1.60	1.71	1.88	2.30	2.65	2.13
Inflation	1.13	0.79	-0.04	0.91	0.64	-0.05	1.47	1.09	-0.02
Short-term interest rate	0.00	0.00	-0.09	0.00	0.00	-0.08	0.00	0.00	-0.05
Long-term interest rate	0.02	0.03	-0.06	-0.06	-0.07	-0.10	-0.15	-0.15	-0.22
Consumption	0.97	1.28	1.53	0.72	0.92	1.53	1.35	1.80	1.81
Investment	1.90	2.54	2.47	0.74	0.76	3.07	3.12	4.72	3.40
Export	1.59	1.52	2.43	1.21	1.06	2.44	1.84	1.61	2.69
Import	2.09	2.31	1.15	1.73	1.77	1.32	2.59	3.16	1.52
Labor	2.75	2.47	0.16	2.21	1.78	0.18	3.36	3.18	0.25
Real wage	0.10	0.46	1.45	0.15	0.53	1.48	0.09	0.49	1.63
Real exch. rate Home-REA	0.02	0.05	0.05	0.02	0.04	0.04	0.03	0.06	0.05
Real exch. rate Home-RW	0.70	0.38	0.44	0.56	0.29	0.43	0.83	0.40	0.47
Public deficit	0.34	0.28	-0.25	-0.63	-0.68	-0.00	-0.00	-0.16	-0.30
Public debt	-3.43	-4.59	-0.89	-2.63	-4.29	-3.67	-2.79	-4.75	-3.62
\mathbf{REA}									
GDP	1.84	1.93	1.62	1.50	1.48	1.69	2.19	2.34	1.90
Inflation	1.32	0.87	-0.05	1.07	0.71	-0.06	1.71	1.20	-0.04
Long-term interest rate	0.04	0.05	-0.05	-0.05	-0.05	-0.09	-0.12	-0.13	-0.21
Real exch. rate REA-RW	3.43	1.54	1.84	2.71	1.13	1.84	4.03	1.52	1.98
Public deficit	0.44	0.41	-0.29	-0.53	-0.55	-0.02	0.15	0.05	-0.32
Public debt	-2.63	-3.33	0.37	-2.11	-3.44	-2.43	-2.18	-3.56	-1.90

Table 8: EA public investment with distortionary taxes and monetary accommodation.

Note: EA public investment increase is a 1% (of EA steady-state GDP) increase in EA-wide public investment. FG represents a EA monetary policy rate kept at its baseline level during the first two years. Tax increase is a symmetric increase of labor, capital, and consumption tax rates whose combined fiscal revenue corresponds to 1% of initial GDP. Monetary policy accommodation is the permanent roll-over of purchases of sovereign bonds by the central bank capable of maintaining the balance sheet of the central bank equal to the total increase in public investment; interest income received by the central bank is transferred back to the government. Average values are reported. Ist= 1st year; 2nd= 2nd year; LR=long run (10 year). REA=rest of the EA. All variables as % dev. from steady state; inflation, interest rate, public deficit/GDP and public debt/(4*GDP) as % point dev. from steady state. For real exchange rates, +=depreciation.

Table 9: EA public investment and PSPP: no-public capital accumulation and time-to-build.

	FG and PSPP		No-public capital acc.			Time-to-build			
	1st	2nd	LR	1st	2nd	LR	1st	2nd	LR
Home									
GDP	3.03	3.55	2.63	1.89	1.94	0.54	1.53	1.86	2.22
Inflation	2.14	1.65	-0.06	1.32	1.12	-0.02	1.51	1.34	-0.15
Short-term interest rate	0.00	0.00	-0.04	-0.00	-0.00	0.06	0.00	0.00	0.10
Long-term interest rate	-0.45	-0.45	-0.34	-0.35	-0.34	-0.19	-0.36	-0.35	-0.24
Consumption	1.96	2.59	2.09	0.92	1.04	0.21	1.29	1.61	1.37
Investment	5.40	8.63	4.45	3.78	5.93	1.17	4.26	6.55	3.76
Export	2.53	2.11	3.46	1.14	0.46	0.73	1.35	0.74	2.92
Import	3.40	4.43	1.72	2.66	3.32	0.33	1.73	2.81	1.54
Labor	4.56	4.50	0.33	3.06	2.91	0.13	2.50	2.78	0.35
Real wage	0.09	0.57	1.99	-0.04	0.10	0.36	-0.03	0.16	1.55
Real exch. rate Home-REA	0.04	0.09	0.06	0.02	0.04	0.02	0.02	0.05	0.05
Real exch. rate Home-RW	1.17	0.53	0.64	0.54	0.13	0.15	0.70	0.22	0.53
Public deficit	-0.61	-0.83	-0.08	-0.09	-0.13	-0.04	-0.89	-0.88	0.08
Public debt	-1.67	-5.32	-6.68	-0.25	-2.11	-1.54	-0.29	-3.59	-3.74
REA									
GDP	2.92	3.15	2.34	1.82	1.71	0.48	1.50	1.66	2.06
Inflation	2.49	1.80	-0.07	1.49	1.19	-0.02	1.70	1.40	-0.12
Long-term interest rate	-0.42	-0.41	-0.32	-0.32	-0.31	-0.18	-0.33	-0.32	-0.23
Real exch. rate REA-RW	5.68	1.96	2.71	2.59	0.28	0.62	3.42	0.73	2.35
Public deficit	-0.37	-0.51	-0.14	0.10	0.09	-0.11	-0.74	-0.66	0.01
Public debt	-1.45	-4.20	-4.11	-0.25	-1.55	-0.06	-0.48	-3.05	-1.79

Note: EA public investment increase is a 1% (of EA steady-state GDP) increase in EA-wide public investment. FG is a EA monetary policy rate kept at its baseline level during the first two years. PSPP is the implementation of the Public Sector Purchase Programme. No-public capital accumulation is an increase in public spending without the corresponding accumulation of public capital. Time-to-build is a delay of five years before the transformation of public investment into accumulation of public capital. Average values are reported. 1st= 1st year; 2nd= 2nd year; LR=long run (10 year). REA=rest of the EA. All variables as % dev. from steady state; inflation, interest rate, public deficit/GDP and public debt/(4*GDP) as % point dev. from steady state. For real exchange rates, +=depreciation.



Figure 1: Home public investment and FG. Home variables

Notes: horizontal axis, quarters; vertical axis, % deviations from the baseline. GDP and its components are both evaluated at constant (steady-state) prices. Inflation and interest rates are reported in annualized p.p., public deficit as ratio to GDP, public debt as a ratio to annualized GDP (all as deviations from the baseline).



Figure 2: Home public investment and FG. REA variables.

Notes: horizontal axis, quarters; vertical axis, % deviations from the baseline. GDP and its components are both evaluated at constant (steady-state) prices. Inflation and interest rates are reported in annualized p.p., public deficit as ratio to GDP, public debt as a ratio to annualized GDP (all as deviations from the baseline).



Figure 3: EA public investment and FG. Home variables.

Notes: horizontal axis, quarters; vertical axis, % deviations from the baseline. GDP and its components are both evaluated at constant (steady-state) prices. Inflation and interest rates are reported in annualized p.p., public deficit as ratio to GDP, public debt as a ratio to annualized GDP (all as deviations from the baseline).



Figure 4: EA public investment and FG. REA variables.

Notes: horizontal axis, quarters; vertical axis, % deviations from the baseline. GDP and its components are both evaluated at constant (steady-state) prices. Inflation and interest rates are reported in annualized p.p., public deficit as ratio to GDP, public debt as a ratio to annualized GDP (all as deviations from the baseline).



Notes: horizontal axis, quarters; vertical axis, % deviations from the baseline. GDP and its components are both evaluated at constant (steady-state) prices. Inflation and interest rates are reported in annualized p.p., public deficit as ratio to GDP, public debt as a ratio to annualized GDP (all as deviations from the baseline).



Figure 6: EA public investment, FG and PSPP. REA variables.

Notes: horizontal axis, quarters; vertical axis, % deviations from the baseline. GDP and its components are both evaluated at constant (steady-state) prices. Inflation and interest rates are reported in annualized p.p., public deficit as ratio to GDP, public debt as a ratio to annualized GDP (all as deviations from the baseline).



Figure 7: Home public investment and spread. Home variables.

Notes: horizontal axis, quarters; vertical axis, % deviations from the baseline. GDP and its components are both evaluated at constant (steady-state) prices. Inflation and interest rates are reported in annualized p.p., public deficit as ratio to GDP, public debt as a ratio to annualized GDP (all as deviations from the baseline).

RECENTLY PUBLISHED "TEMI" (*)

- N. 1129 *The effects of central bank's verbal guidance: evidence from the ECB*, by Maddalena Galardo and Cinzia Guerrieri (July 2017).
- N. 1130 The Bank of Italy econometric model: an update of the main equations and model elasticities, by Guido Bulligan, Fabio Busetti, Michele Caivano, Pietro Cova, Davide Fantino, Alberto Locarno, Lisa Rodano (July 2017).
- N.1131 Venture capitalists at work: what are the effects on the firms they finance?, by Raffaello Bronzini, Giampaolo Caramellino and Silvia Magri (September 2017).
- N.1132 Low frequency drivers of the real interest rate: a band spectrum regression approach, by Fabio Busetti and Michele Caivano (September 2017).
- N. 1133 *The real effects of relationship lending*, by Ryan Banerjee, Leonardo Gambacorta and Enrico Sette (September 2017).
- N.1134 *Credit demand and supply: a two-way feedback relation*, by Ugo Albertazzi and Lucia Esposito (September 2017).
- N. 1135 Legislators' behaviour and electoral rules: evidence from an Italian reform, by Giuseppe Albanese, Marika Cioffi and Pietro Tommasino (September 2017).
- N.1136 Macroeconomic effects of non-standard monetary policy measures in the euro area: the role of corporate bond purchases, by Anna Bartocci, Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (September 2017).
- N. 1137 On secular stagnation and low interest rates: demography matters, by Giuseppe Ferrero, Marco Gross and Stefano Neri (September 2017).
- N. 1138 Tony Atkinson and his legacy, by Rolf Aaberge, François Bourguignon, Andrea Brandolini, Francisco H. G. Ferreira, Janet C. Gornick, John Hills, Markus Jäntti, Stephen P. Jenkins, Eric Marlier, John Micklewright, Brian Nolan, Thomas Piketty, Walter J. Radermacher, Timothy M. Smeeding, Nicholas H. Stern, Joseph Stiglitz and Holly Sutherland (September 2017).
- N. 1139 *Credit misallocation during the European financial crisis*, by Fabiano Schivardi, Enrico Sette and Guido Tabellini (September 2017).
- N. 1140 Natural rates across the Atlantic, by Andrea Gerali and Stefano Neri (September 2017).
- N.1141 A quantitative analysis of risk premia in the corporate bond market, by Sara Cecchetti (October 2017).
- N. 1142 *Monetary policy in times of debt*, by Mario Pietrunti and Federico M. Signoretti (October 2017).
- N.1143 Capital misallocation and financial development: a sector-level analysis, by Daniela Marconi and Christian Upper (October 2017).
- N. 1144 Leaving your mamma: why so late in Italy?, by Enrica Di Stefano (October 2017).
- N. 1145 A Financial Conditions Index for the CEE economies, by Simone Auer (October 2017).
- N. 1146 Internal capital markets in times of crisis: the benefit of group affiliation in Italy, by Raffaele Santioni, Fabio Schiantarelli and Philip E. Strahan (October 2017).
- N. 1147 *Consistent inference in fixed-effects stochastic frontier models*, by Federico Belotti and Giuseppe Ilardi (October 2017).
- N. 1148 Investment decisions by European firms and financing constraints, by Andrea Mercatanti, Taneli Mäkinen and Andrea Silvestrini (October 2017).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBERTAZZI U., G. ERAMO, L. GAMBACORTA and C. SALLEO, Asymmetric information in securitization: an empirical assessment, Journal of Monetary Economics, v. 71, pp. 33-49, TD No. 796 (February 2011).
- ALESSANDRI P. and B. NELSON, *Simple banking: profitability and the yield curve,* Journal of Money, Credit and Banking, v. 47, 1, pp. 143-175, **TD No. 945 (January 2014).**
- ANTONIETTI R., R. BRONZINI and G. CAINELLI, *Inward greenfield FDI and innovation*, Economia e Politica Industriale, v. 42, 1, pp. 93-116, **TD No. 1006 (March 2015).**
- BARONE G. and G. NARCISO, Organized crime and business subsidies: Where does the money go?, Journal of Urban Economics, v. 86, pp. 98-110, **TD No. 916 (June 2013).**
- BRONZINI R., The effects of extensive and intensive margins of FDI on domestic employment: microeconomic evidence from Italy, B.E. Journal of Economic Analysis & Policy, v. 15, 4, pp. 2079-2109, TD No. 769 (July 2010).
- BUGAMELLI M., S. FABIANI and E. SETTE, The age of the dragon: the effect of imports from China on firmlevel prices, Journal of Money, Credit and Banking, v. 47, 6, pp. 1091-1118, TD No. 737 (January 2010).
- BULLIGAN G., M. MARCELLINO and F. VENDITTI, *Forecasting economic activity with targeted predictors,* International Journal of Forecasting, v. 31, 1, pp. 188-206, **TD No. 847 (February 2012).**
- BUSETTI F., On detecting end-of-sample instabilities, in S.J. Koopman, N. Shepard (eds.), Unobserved Components and Time Series Econometrics, Oxford, Oxford University Press, **TD No. 881 (September 2012).**
- CESARONI T., *Procyclicality of credit rating systems: how to manage it,* Journal of Economics and Business, v. 82. pp. 62-83, **TD No. 1034 (October 2015).**
- CIARLONE A., *House price cycles in emerging economies*, Studies in Economics and Finance, v. 32, 1, **TD No. 863 (May 2012).**
- CUCINIELLO V. and F. M. SIGNORETTI, *Large bank, loan rate markup and monetary policy,* International Journal of Central Banking, v. 11, 3, pp. 141-177, **TD No. 987 (November 2014).**
- DE BLASIO G., D. FANTINO and G. PELLEGRINI, *Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds,* Industrial and Corporate Change, v. 24, 6, pp. 1285-1314, **TD No. 792 (February 2011).**
- DEPALO D., R. GIORDANO and E. PAPAPETROU, *Public-private wage differentials in euro area countries:* evidence from quantile decomposition analysis, Empirical Economics, v. 49, 3, pp. 985-1115, **TD No. 907 (April 2013).**
- DI CESARE A., A. P. STORK and C. DE VRIES, *Risk measures for autocorrelated hedge fund returns*, Journal of Financial Econometrics, v. 13, 4, pp. 868-895, **TD No. 831 (October 2011).**
- FANTINO D., A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, v. 1, 2, pp. 219-251, TD No. 884 (October 2012).
- FRATZSCHER M., D. RIMEC, L. SARNOB and G. ZINNA, *The scapegoat theory of exchange rates: the first tests*, Journal of Monetary Economics, v. 70, 1, pp. 1-21, **TD No. 991 (November 2014).**
- NOTARPIETRO A. and S. SIVIERO, *Optimal monetary policy rules and house prices: the role of financial frictions*, Journal of Money, Credit and Banking, v. 47, S1, pp. 383-410, **TD No. 993 (November 2014).**
- RIGGI M. and F. VENDITTI, *The time varying effect of oil price shocks on euro-area exports,* Journal of Economic Dynamics and Control, v. 59, pp. 75-94, **TD No. 1035 (October 2015).**
- TANELI M. and B. OHL, *Information acquisition and learning from prices over the business cycle,* Journal of Economic Theory, 158 B, pp. 585–633, **TD No. 946 (January 2014).**

- ALBANESE G., G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **TD No. 955 (March 2014).**
- ANDINI M. and G. DE BLASIO, *Local development that money cannot buy: Italy's Contratti di Programma,* Journal of Economic Geography, v. 16, 2, pp. 365-393, **TD No. 915 (June 2013).**
- BARONE G. and S. MOCETTI, *Inequality and trust: new evidence from panel data*, Economic Inquiry, v. 54, pp. 794-809, **TD No. 973 (October 2014).**
- BELTRATTI A., B. BORTOLOTTI and M. CACCAVAIO, *Stock market efficiency in China: evidence from the split-share reform*, Quarterly Review of Economics and Finance, v. 60, pp. 125-137, **TD No. 969** (October 2014).
- BOLATTO S. and M. SBRACIA, *Deconstructing the gains from trade: selection of industries vs reallocation of workers*, Review of International Economics, v. 24, 2, pp. 344-363, **TD No. 1037 (November 2015).**
- BOLTON P., X. FREIXAS, L. GAMBACORTA and P. E. MISTRULLI, *Relationship and transaction lending in a crisis*, Review of Financial Studies, v. 29, 10, pp. 2643-2676, **TD No. 917 (July 2013).**
- BONACCORSI DI PATTI E. and E. SETTE, Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register, Journal of Financial Intermediation, v. 25, 1, pp. 54-76, TD No. 848 (February 2012).
- BORIN A. and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, v. 152, 4, pp. 705-732, **TD No. 1011 (June 2015).**
- BRAGOLI D., M. RIGON and F. ZANETTI, *Optimal inflation weights in the euro area*, International Journal of Central Banking, v. 12, 2, pp. 357-383, **TD No. 1045 (January 2016).**
- BRANDOLINI A. and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, **TD No. 965 (July 2015).**
- BRIPI F., *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, v. 30, 2, pp. 383-411, **TD No. 932 (September 2013).**
- BRONZINI R. and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **TD No. 960 (April 2014).**
- BURLON L. and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, **TD No. 963 (June 2014).**
- BUSETTI F. and M. CAIVANO, *The trend-cycle decomposition of output and the Phillips Curve: bayesian estimates for Italy and the Euro Area,* Empirical Economics, V. 50, 4, pp. 1565-1587, **TD No. 941** (November 2013).
- CAIVANO M. and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, **TD No. 947 (January 2014).**
- CALZA A. and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **TD No. 1039 (December 2015).**
- CESARONI T. and R. DE SANTIS, *Current account "core-periphery dualism" in the EMU*, The World Economy, v. 39, 10, pp. 1514-1538, **TD No. 996 (December 2014).**
- CIANI E., *Retirement, Pension eligibility and home production,* Labour Economics, v. 38, pp. 106-120, **TD** No. 1056 (March 2016).
- CIARLONE A. and V. MICELI, Escaping financial crises? Macro evidence from sovereign wealth funds' investment behaviour, Emerging Markets Review, v. 27, 2, pp. 169-196, TD No. 972 (October 2014).
- CORNELI F. and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, v. 65, pp. 166-194, **TD No. 1012 (June 2015).**
- D'AURIZIO L. and D. DEPALO, An evaluation of the policies on repayment of government's trade debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **TD No. 1061 (April 2016).**
- DE BLASIO G., G. MAGIO and C. MENON, Down and out in Italian towns: measuring the impact of economic downturns on crime, Economics Letters, 146, pp. 99-102, TD No. 925 (July 2013).
- DOTTORI D. and M. MANNA, *Strategy and tactics in public debt management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **TD No. 1005 (March 2015).**
- ESPOSITO L., A. NOBILI and T. ROPELE, *The management of interest rate risk during the crisis: evidence from Italian banks*, Journal of Banking & Finance, v. 59, pp. 486-504, **TD No. 933 (September 2013).**

- LIBERATI D., M. MARINUCCI and G. M. TANZI, Science and technology parks in Italy: main features and analysis of their effects on hosted firms, Journal of Technology Transfer, v. 41, 4, pp. 694-729, TD No. 983 (November 2014).
- MARCELLINO M., M. PORQUEDDU and F. VENDITTI, *Short-Term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility,* Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, **TD No. 896 (January 2013).**
- RODANO G., N. SERRANO-VELARDE and E. TARANTINO, *Bankruptcy law and bank financing*, Journal of Financial Economics, v. 120, 2, pp. 363-382, **TD No. 1013 (June 2015).**
- ZINNA G., Price pressures on UK real rates: an empirical investigation, Review of Finance, v. 20, 4, pp. 1587-1630, TD No. 968 (July 2014).

2017

- ADAMOPOULOU A. and G.M. TANZI, Academic dropout and the great recession, Journal of Human Capital, V. 11, 1, pp. 35–71, **TD No. 970 (October 2014).**
- ALBERTAZZI U., M. BOTTERO and G. SENE, Information externalities in the credit market and the spell of credit rationing, Journal of Financial Intermediation, v. 30, pp. 61–70, TD No. 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **TD No. 977 (November 2014).**
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp.59-72, **TD No. 761 (June 2010).**
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, **TD No. 1085** (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 TD No. 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, **TD No. 1049 (January 2016).**
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **TD No. 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. a model-based evaluation, International Finance, v. 20, 2, pp.155-173, TD No. 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts,* Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **TD No. 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, **TD No. 1031 (October 2015).**
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, TD No. 1018 (July 2015).
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment,* International Journal of Forecasting, v. 33, 4, pp. 801-816, **TD No. 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, **TD No. 953 (March 2014).**
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, TD No. 942 (November 2013).
- DELLE MONACHE D. and I. PETRELLA, Adaptive models and heavy tails with an application to inflation forecasting, International Journal of Forecasting, v. 33, 2, pp. 482-501, TD No. 1052 (March 2016).
- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, TD No. 1108 (April 2017).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **TD No. 1033 (October 2015).**

- MANCINI A.L., C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, v. 15, 3, pp 965–993, **D No. 958** (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **TD No. 939 (November 2013).**
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **TD No. 763 (June 2010).**
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, **TD No. 752 (March 2010)**.
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **TD No. 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, A structural model for the housing and credit market in Italy, Journal of Housing Economics, v. 36, pp. 73-87, **TD No. 887 (October 2012).**
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, **TD No. 1073 (July 2016).**
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, TD No. 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, **TD No. 1048 (January 2016).**
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach,* International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **TD No. 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, v. 30, 10, pp. 3538–3580, **TD No. 1065 (April 2016).**
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, TD No. 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, **TD No. 734 (November 2009).**
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, **TD No. 1104 (February 2017).**

FORTHCOMING

- ADAMOPOULOU A. and E. KAYA, Young Adults living with their parents and the influence of peers, Oxford Bulletin of Economics and Statistics, **TD No. 1038 (November 2015).**
- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **TD No. 911 (April 2013).**
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis,* Journal of the European Economic Association, **TD No. 909 (April 2013).**
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, **TD No. 1077 (July 2016).**
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, TD No. 787 (January 2011).
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **TD No. 877 (September 2012).**
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, **TD No. 898 (January 2013).**
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, **TD No. 1025 (July 2015).**
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, **TD No. 871 (July 2012).**
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, TD No. 1100 (February 2017).